NHS SUPPLY CHAIN FRAMEWORK AGREEMENT FOR THE SUPPLY OF GOODS

NHS Supply Chain	Supply Chain Coordination Limited (registered number 10881715) whose registered office is at Wellington House, 133-155 Waterloo Road, London, United Kingdom, SE1 8UG;
The Supplier	[Insert company name, registered number and registered office address]
Date	
Type of Goods	Supplies
Framework Agreement Name	Airway Management Products and Associated Equipment
Framework Agreement Number	Project_1008
Lot	Lot 1 - Laryngoscopes, Video Laryngoscopes, Tracheal Intubation Equipment, Single Use Upper Airway Devices, Single use Bronchoscopes and Single Use Flexible Intubating Endoscopes and Related Accessories
	Lot 2 - Endotracheal Tubes, Endobronchial Tubes and Blockers, Tracheostomy Tubes, Supraglottic Airways and Simple Airway Adjuncts
	Lot 3 - Breathing Systems Circuits and Accessories
	Lot 4 - Humidification and Filtration (HME's, HMEF's, Bacterial and Viral Filters)

This Framework Agreement is made on the date set out above subject to the terms set out in the schedules listed below ("**Schedules**"). NHS Supply Chain (operated by Supply Chain Coordination Limited) and the Supplier undertake to comply with the provisions of the Schedules in the performance of this Framework Agreement.

The Definitions in Schedule 4 apply to the use of all capitalised terms in this Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 1 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedules

Schedule 1	Key Provisions	
Schedule 2	General Terms and Conditions	
Schedule 3	Information Governance Provisions	
Schedule 4	Definitions and Interpretations	
Schedule 5(a)	Specification	
Schedule 5(b)	Tender Response Document	
Schedule 6	Commercial Schedule	
Schedule 7	Ordering Procedure and Order Form	
Schedule 8 Service Levels		
Schedule 9	Call-off Terms and Conditions for the Supply of Goods	

Signed by an authorised representative for and on behalf of NHS Supply Chain operated by Supply Chain Coordination Limited

Name:	 Signature:	
Position:		

Signed by an authorised representative of the Supplier

Name:	 	Signature	
Position:	 		

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 2 of 9		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 1

Key Provisions

Standard Key Provisions

1 Application of the Key Provisions

- 1.1 The standard Key Provisions at Clauses 1 to 7 of this Schedule 1 shall apply to this Framework Agreement.
- 1.2 The optional Key Provisions at Clauses 8 to 10 of this Schedule 1 shall only apply to this Framework Agreement where they have been checked and information completed as applicable.
- 1.3 Extra Key Provisions shall only apply to this Framework Agreement where such provisions are set out at the end of this Schedule 1.

2 Term

2.1 The Term of this Framework Agreement shall be twenty-four months from the Commencement Date and may be extended in accordance with Clause 16.2 of Schedule 2 provided that the duration of this Framework Agreement shall be no longer than forty-eight months in total.

3 Contract Managers

- 3.1 The Contract Managers at the commencement of this Framework Agreement are:
 - 3.1.1 for NHS Supply Chain:

Buyer - Airway Management Products and Associated Equipment

3.1.2 for the Supplier:

Contract Manager.

4 Names and addresses for notices

- 4.1 Notices served under this Framework Agreement are to be delivered to:
 - 4.1.1 for NHS Supply Chain:

Procurement Director, Foxbridge Way, Normanton, WF6 1TL

4.1.2 for the Supplier:

Director, [address]

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 3 c		Page 3 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Commented [SJ1]: Will this still be PD and which address we using?

5 Management levels for escalation and dispute resolution

5.1 The management levels at which a dispute will be dealt with are as follows:

Level	NHS Supply Chain representative	Supplier representative
1	Buyer	Contract Manager or equivalent
2	Senior Buyer	Senior Contract Manager or equivalent
3	Head of Category	Assistant Director or equivalent
4	Procurement Director	Director or equivalent

6 Order of precedence

- 6.1 Subject always to Clause 1.10 of Schedule 4, should there be a conflict between any other parts of this Framework Agreement the order of priority for construction purposes shall be:
 - 6.1.1 the provisions on the front page of this NHS Framework Agreement for the Supply of Goods;
 - 6.1.2 Schedule 1: Key Provisions;
 - 6.1.3 Schedule 5(a): Specification;
 - 6.1.4 Schedule 2: General Terms and Conditions;
 - 6.1.5 Schedule 6: Commercial Schedule;
 - 6.1.6 Schedule 8: Service Levels;
 - 6.1.7 Schedule 5(b): Tender Response Document;
 - 6.1.8 Schedule 3: Information Governance Provisions;
 - 6.1.9 Schedule 4: Definitions and Interpretations;
 - 6.1.10 the order in which all subsequent schedules, if any, appear; and
 - 6.1.11 any other documentation forming part of the Framework Agreement in the date order in which such documentation was created with the more recent documentation taking precedence over older documentation to the extent only of any conflict.

7 Participating Authorities

7.1 The following Contracting Authorities are entitled to place Orders:

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 4 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 7.1.1 in relation to a Direct Route of Supply: any NHS Trust; other NHS entities; any private sector entity which is active in the United Kingdom Healthcare Sector; or any government department, government agency or other statutory body; and
- 7.1.2 in relation to a Non-direct Route of Supply: NHS Supply Chain,

for the avoidance of doubt, any successor bodies of any of the entities described in this definition are included in this definition.

Optional Key Provisions

- 8 Quality assurance standards self-certification [] (only applicable to the Framework Agreement if this box is checked and the standards are listed)
- 8.1 The Supplier warrants that on the request of NHS Supply Chain it shall provide a written and signed self-certification in the form requested by NHS Supply Chain that it complies, and will notify NHS Supply Chain immediately if it no longer complies throughout the Term of the Framework Agreement and all Contracts with all quality assurance standards applicable to the Goods and Services and that it shall evidence such compliance on request.
- 9 Different levels and/or types of insurance [] (only applicable to the Framework Agreement if this box is checked and the table sets out the requirements)
- 9.1 The Supplier shall put in place and maintain in force the following insurances with the following minimum cover per claim:

Type of insurance required	Minimum cover

10 Guarantee (only applicable to the Framework Agreement if this box is checked)

10.1 Promptly following the execution of this Framework Agreement, the Supplier shall, if it has not already delivered an executed deed of guarantee to NHS Supply Chain, deliver the executed deed of guarantee to NHS Supply Chain as required by the procurement process followed by NHS Supply Chain. Failure to comply with this Key Provision shall be an irremediable breach of this Framework Agreement.

Extra Key Provisions

11.1 Suppliers shall be aware that they will be required to adopt and apply PPN 06/20 'Taking Account of Social Value in the Award of Central Government Contracts' and support NHS Supply Chain to deliver social value objectives as evidenced in a number of ways including the following:

	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 5 of 5		Page 5 of 90	
	NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Commented [SJ2]: Wasn't everything about Direct Route being removed?

- 11.1.1 To support working towards net zero greenhouse gas emissions and effective stewardship of the Environment through the lifetime of the Framework Agreement, within six (6) months of the Commencement Date the Supplier shall provide NHS Supply Chain with evidence (to the absolute satisfaction of NHS Supply Chain) that they have started to assess and map the environmental impacts of their Organisation. The Parties agree that where the Supplier does not provide this evidence to NHS Supply Chain within six (6) months of the Commencement Date this shall be deemed a material breach for the purposes of Clause 16.4 of Schedule 2 below, meaning that NHS Supply Chain reserves the right (acting in it absolute discretion) to use any of the remedial rights set out in the Framework Agreement, including but not limited to those set out at Clauses 16, 17 and 18 of Schedule 2 above.
- 11.1.2 To support working towards net zero greenhouse gas emissions and effective stewardship of the Environment through the lifetime of the Framework Agreement, within eighteen (18) months of the Commencement Date the Supplier shall provide NHS Supply Chain with evidence (to the absolute satisfaction of NHS Supply Chain) of their formal commitment the Organisation is making to reduce their environmental impact arising from the performance of the contract and to continuous improvement year on year through the lifetime of the Framework Agreement. The Parties agree that where the Supplier does not provide this evidence to NHS Supply Chain within eighteen (18) months of the Commencement Date this shall be deemed a material breach for the purposes of Clause 16.4 of Schedule 2 below, meaning that NHS Supply Chain reserves the right (acting in it absolute discretion) to use any of the remedial rights set out in the Framework Agreement, including but not limited to those set out at Clauses 16, 17 and 18 of Schedule 2 above.
- 11.1.3 In line with Social Value Priority Theme 3 'Fighting Climate Change', and the published reporting metrics within PPN06/20, every quarter the Supplier shall provide NHS Supply Chain with an update on their progress towards achieving their anticipated metrics in relation to: the number of green spaces created under the contract; the annual reduction in emissions of greenhouse gases arising from the performance of the contract, measured in metric tonnes carbon dioxide equivalents (MTCDE); the annual reduction in water use arising from the performance of the contract, measured in litres; and the annual reduction in waste to landfill arising from the performance of the contract, measured in the contract, measured in metric tonnes.
- 11.2 NHS Supply Chain have adopted the Supplier Registration Systems for the Government (SRS) for both Supplier Assessment and Supplier Mapping, the purpose being to increase oversight and enable due diligence on Sustainability and Social Value aspects. This is to support NHS Supply Chain's Sustainability strategy commitments; both to support the NHS's ambition to decarbonise and journey to net zero ('Delivering a Net Zero NHS') and to understand what due diligence suppliers have in place to manage compliance with international labour standards:
 - 11.2.1 Within six (6) months of the Commencement Date the Supplier is required to complete the Labour Standards Assessment and Carbon Waste and Water Assessment Tools (the template for which can be found at the following link: https://supplierregistration.cabinetoffice.gov.uk/). The Parties agree that where the Supplier fails to complete the Assessment Tools within

	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Pag		Page 6 of 90	
	NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

six (6) months of the Commencement Date this shall be deemed a material breach for the purposes of Clause 16.4 of Schedule 2 below, meaning that NHS Supply Chain may use any of the remedial rights set out in the Framework Agreement, including but not limited to those set out at Clauses 16, 17 and 18 of Schedule 2 below.

- 11.2.2 Within three (3) months of the Commencement Date the Supplier shall provide NHS Supply Chain with a completed version of the Modern Slavery Assessment Tool (the template for which can be found at the following link: https://supplierregistration.cabinetoffice.gov.uk/) and the Parties agree that where the Supplier does not provide a completed version of this Assessment Tool to NHS Supply Chain within three (3) months of the Commencement Date this shall be deemed a material breach for the purposes of Clause 16.4 of Schedule 2 below, meaning that NHS Supply Chain may use any of the remedial rights set out in the Framework Agreement, including but not limited to those set out at Clauses 16, 17 and 18 of Schedule 2 below.
- 11.3 To support NHS Supply Chain in delivering the recommendations made following The Boardman Review into Pandemic Procurement and particularly the requirement to increase transparency of Supply Chains to the NHS:
 - 11.3.1 Within six (6) months of the Commencement Date, NHS Supply Chain requires Suppliers, to complete Supply Chain Mapping for all products on the Framework Agreement using the Supply Chain Mapping Tool (which can be found at the following link: https://supplierregistration.cabinetoffice.gov.uk/).

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017 Page 7 of 90	
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 2

General Terms and Conditions

	itents
1.	Supplier's appointment
2.	NHS Supply Chain commitments
3.	Ordering procedures
4.	Reasonable assistance
5.	Supplier performance
6.	Business continuity
	NHS Supply Chain's obligations
	Contract management
	Price and payment
	Warranties
11.	Intellectual Property
12.	Statutory compliance
13.	Independence of Participating Authorities
14.	Limitation of liability
15.	Insurance
16.	Term and termination
17.	Consequences of expiry or earlier termination of this Framework Agreement
18.	Suspension of Supplier's appointment
19.	Complaints process
	Sustainable development
21.	Electronic product information
	Change management
23.	Dispute resolution
	Force majeure
25.	Records retention and right of audit
	Conflicts of interest and the prevention of fraud
27.	Equality and human rights
28.	Notice
	Assignment, novation and Sub-contracting
30.	Prohibited Acts
31.	General

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 8 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

1 <u>Supplier's appointment</u>

- 1.1 NHS Supply Chain appoints the Supplier as a potential supplier of the Goods and Services and the Supplier shall be eligible to be considered for the award of Orders during the Term.
- 1.2 In consideration of NHS Supply Chain agreeing to appoint the Supplier to this Framework Agreement in accordance with Clause 1.1 of this Schedule 2 and the mutual exchange of promises and obligations under this Framework Agreement, the Supplier undertakes to supply Goods and Services under Orders placed with the Supplier:
 - 1.2.1 of the exact quality, type and as otherwise specified in the Specification and accepted by NHS Supply Chain in the Tender Response Document;
 - 1.2.2 at the Contract Price calculated in accordance with the Commercial Schedule; and
 - 1.2.3 in such quantities, at such times and to such locations as may be specified in an Order.
- 1.3 The Supplier agrees that the Call-Off Terms and Conditions for the Supply of Goods shall apply to all supplies of the Goods and any associated Services made by the Supplier to a Participating Authority pursuant to this Framework Agreement. The Supplier agrees that it will not in its dealings with a Participating Authority seek to impose or rely on any other contractual terms which in any way vary or contradict the relevant Contract.
- 1.4 The Supplier shall comply fully with its obligations set out in this Framework Agreement, the Specification and Tender Response Document, the Call-off Terms and Conditions for the Supply of Goods and any other provisions of Contracts entered into under and in accordance with this Framework Agreement (to include, without limitation, the KPIs and all obligations in relation to the quality, performance characteristics, supply and delivery in relation to use of the Goods and the provision of the Services).
- 1.5 Without limitation to any of the provisions of Clause 22 of this Schedule 2 and/or the Commercial Schedule:
 - 1.5.1 The Supplier agrees to work with NHS Supply Chain during the Term of this Framework Agreement to achieve continuous and innovative improvements to the quality and value of the Goods and Services, including the way in which the Goods and Services are sourced, supplied, ordered and packaged, to achieve the most efficient and best value Goods and Services for the mutual benefit of the Supplier, NHS Supply Chain, the Authority and NHS.
 - 1.5.2 The Supplier agrees to work with NHS Supply Chain during the Term of this Framework Agreement to explore ways in which commitment offered by Authorities in relation to specific Contracts can be reflected in more competitive pricing for the Authority.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 9 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 1.6 If there are any quality, performance and/or safety related reports, notices, alerts or other communications issued by the Supplier or any regulatory or other body in relation to the Goods, the Supplier shall promptly (which in the case of any incidents which may have an effect on patient safety, shall mean within one (1) Business Day) provide NHS Supply Chain with a copy of any such reports, notices, alerts or other communications.
- 1.7 Upon receipt of any such reports, notices, alerts or other communications pursuant to Clause 1.6 of this Schedule 2, NHS Supply Chain shall be entitled to request further information from the Supplier and/or a meeting with the Supplier, and the Supplier shall cooperate fully in all matters relating to any such request.
- 1.8 In complying with its obligations under this Framework Agreement, the Supplier shall, and shall procure that all Staff shall, act in accordance with the NHS values as set out in the NHS Constitution from time to time.

2 NHS Supply Chain commitments

- 2.1 Unless otherwise set out in the Commercial Schedule, the Supplier acknowledges that:
 - 2.1.1 there is no obligation for NHS Supply Chain or for any other Participating Authority to purchase any Goods or Services from the Supplier during the Term;
 - 2.1.2 no undertaking or any form of statement, promise, representation or obligation has been made by NHS Supply Chain and/or any other Participating Authority in respect of the total quantities or value of the Goods and/or Services to be ordered by them pursuant to this Framework Agreement and the Supplier acknowledges and agrees that it has not entered into this Framework Agreement on the basis of any such undertaking, statement, promise or representation;
 - 2.1.3 in entering this Framework Agreement, no form of exclusivity has been granted by NHS Supply Chain and/or any other Participating Authority; and
 - 2.1.4 NHS Supply Chain and/or other Participating Authorities are at all times entitled to enter into other contracts and agreements with other suppliers for the provision of any or all goods and services which are the same as or similar to the Goods and Services.

3 Ordering procedure

3.1 Any Participating Authority may enter into Contracts by placing an Order in accordance with the Ordering Procedure.

4 Reasonable assistance

4.1 Upon the written request of any Participating Authority, the Supplier shall provide such Participating Authority with any reasonable and proportionate information that it holds about the Goods and Services it supplies under this Framework Agreement including, without limitation, the compatibility and interoperability of the Goods with other products, to enable the Participating Authority to complete any necessary due diligence before purchasing such Goods and/or Services.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 10 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

5 <u>Supplier performance</u>

- 5.1 The Supplier shall perform all Contracts entered into under this Framework Agreement by NHS Supply Chain or any other Participating Authority in accordance with:
 - 5.1.1 the requirements of this Framework Agreement; and
 - 5.1.2 the provisions of the respective Contracts.

6 Business continuity

- 6.1 Throughout the Term, the Supplier will ensure its Business Continuity Plan provides for continuity during a Business Continuity Event. The Supplier confirms and agrees such Business Continuity Plan details and will continue to detail robust arrangements that are reasonable and proportionate to:
 - 6.1.1 the criticality of this Framework Agreement to the Participating Authorities; and
 - 6.1.2 the size and scope of the Supplier's business operations,

regarding continuity of the supply of Goods and Services during and following a Business Continuity Event.

- 6.2 The Supplier shall test its Business Continuity Plan at reasonable intervals, and in any event no less than once every twelve (12) months or such other period as may be agreed between the Parties taking into account the criticality of this Framework Agreement to Participating Authorities and the size and scope of the Supplier's business operations. The Supplier shall promptly provide to NHS Supply Chain, at NHS Supply Chain's written request, copies of its Business Continuity Plan, reasonable and proportionate documentary evidence that the Supplier tests its Business Continuity Plan in accordance with the requirements of this Clause 6.2 of this Schedule 2 and reasonable and proportionate information regarding the outcome of such tests. The Supplier shall provide to NHS Supply Chain a copy of any updated or revised Business Continuity Plan within fourteen (14) Business Days of any material update or revision to the Business Continuity Plan.
- 6.3 NHS Supply Chain may suggest reasonable and proportionate amendments to the Supplier regarding the Business Continuity Plan at any time. Where the Supplier, acting reasonably, deems such suggestions made by NHS Supply Chain to be relevant and appropriate, the Supplier will incorporate into the Business Continuity Plan all such suggestions made by NHS Supply Chain in respect of such Business Continuity Plan. Should the Supplier not incorporate any suggestion made by NHS Supply Chain into such Business Continuity Plan it will explain the reasons for not doing so to NHS Supply Chain.
- 6.4 Should a Business Continuity Event occur at any time, the Supplier shall implement and comply with its Business Continuity Plan and provide regular written reports to NHS Supply Chain on such implementation.
- 6.5 During and following a Business Continuity Event, the Supplier shall use reasonable endeavours to continue to fulfil its obligations in accordance with this Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 11 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

7 NHS Supply Chain's obligations

- 7.1 NHS Supply Chain shall, provide reasonable cooperation to the Supplier as appropriate, provide copies of or give the Supplier access to such of the Policies that are relevant to the Supplier complying with its obligations under this Framework Agreement.
- 7.2 NHS Supply Chain shall comply with NHS Supply Chain's Obligations, if any.

8 Contract management

- 8.1 Each Party shall appoint and retain a Contract Manager who shall be the primary point of contact for the other Party in relation to matters arising from this Framework Agreement. Should the Contract Manager be replaced, the Party replacing the Contract Manager shall promptly inform the other Party in writing of the name and contact details for the new Contract Manager. Any Contract Manager appointed shall be of sufficient seniority and experience to be able to make decisions on the day to day operation of the Framework Agreement. The Supplier confirms and agrees that it will be expected to work closely and cooperate fully with NHS Supply Chain's Contract Manager.
- 8.2 Each Party shall ensure that its representatives (to include, without limitation, its Contract Manager) shall attend review meetings on a regular basis to review the performance of the Supplier under this Framework Agreement and to discuss matters arising generally under this Framework Agreement. Each Party shall ensure that those attending such meetings have the authority to make decisions regarding the day to day operation of the Framework Agreement. Review meetings shall take place at the frequency specified in the Specification. Should the Specification not state the frequency, then meetings shall take place at intervals as may otherwise be agreed in writing between the Parties.
- 8.3 Two weeks prior to each review meeting (or at such time and frequency as may be specified in the Specification) the Supplier shall provide a written contract management report to NHS Supply Chain regarding the supply of the Goods and Services and the operation of this Framework Agreement. Unless otherwise agreed by the Parties in writing, such contract management report shall contain:
 - 8.3.1 details of the performance of the Supplier under this Framework Agreement and any Contracts when assessed in accordance with the KPIs, as relevant to the Framework Agreement and any Contracts, since the last such performance report;
 - 8.3.2 details of any complaints by Participating Authorities in relation to the supply of Goods and Services, their nature and the way in which the Supplier has responded to such complaints since the last review meeting written report;
 - 8.3.3 any information specified in the Specification as being relevant to the operation of this Framework Agreement;
 - 8.3.4 a status report in relation to the implementation of any current Remedial Proposals and Action Plans; and
 - 8.3.5 such other information as reasonably required by NHS Supply Chain.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 12 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

- 8.4 Unless otherwise agreed between the Parties, NHS Supply Chain shall take minutes of each review meeting and shall circulate draft minutes to the Supplier within a reasonable time following such review meeting. The Supplier shall inform NHS Supply Chain in writing of any suggested amendments to the minutes within five (5) Business Days of receipt of the draft minutes. If the Supplier does not respond to NHS Supply Chain within such five (5) Business Days the minutes will be deemed to be approved. Where there are any differences in interpretation of the minutes, the Parties will use their reasonable endeavours to reach agreement. If agreement cannot be reached the matter shall be referred to, and resolved in accordance with, the dispute resolution process set out in Clause 5 of the Key Provisions and Clause 23.2 of this Schedule 2.
- 8.5 The Supplier shall provide any management information required in accordance with the ORS (including, for the avoidance of doubt, monthly statements) and as NHS Supply Chain may request from time to time within seven (7) Business Days of the date of the request. The Supplier shall supply the management information to NHS Supply Chain in such form as may be specified by the ORS or NHS Supply Chain and, where requested to do so, the Supplier shall also provide such management information in accordance with UK government policy (to include, without limitation, for the purposes of analysing public sector expenditure and planning future procurement activities) ("Third Party Body"). The Supplier confirms and agrees that NHS Supply Chain may itself provide the Third Party Body with management information relating to the Goods and Services ordered and any payments made under this Framework Agreement or any Contracts and any other information relevant to the operation of this Framework Agreement.
- 8.6 Upon receipt of management information supplied by the Supplier to NHS Supply Chain and/or the Third Party Body, or by NHS Supply Chain to the Third Party Body, the Parties hereby consent to the Third Party Body and NHS Supply Chain:
 - 8.6.1 storing and analysing the management information and producing statistics; and
 - 8.6.2 sharing the management information or any statistics produced using the management information with any other Contracting Authority.
- 8.7 If the Third Party Body and/or NHS Supply Chain shares the management information or any other information provided under Clause 8.6 of this Schedule 2, any Contracting Authority receiving the management information shall, where such management information is subject to obligations of confidence under this Framework Agreement and such management information is provided direct by NHS Supply Chain to such Contracting Authority, be informed of the confidential nature of that information by NHS Supply Chain and shall be requested by NHS Supply Chain not to disclose it to any body that is not a Contracting Authority (unless required to do so by Law).
- 8.8 NHS Supply Chain may make changes to the type of management information which the Supplier is required to supply and shall give the Supplier at least one (1) month's written notice of any changes.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 13 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

9 Price and payment

Contract Price

9.1 The Contract Price for all Contracts shall be calculated as set out in the Commercial Schedule and the payment provisions for all Contracts shall be as set out in the Calloff Terms and Conditions for the Supply of Goods.

Management Fee

- 9.2 Where Goods and any incidental Services are ordered and delivered via a Direct Route of Supply the Supplier shall pay to NHS Supply Chain a management fee as a percentage of the total Order value at the rate set out in the Commercial Schedule (the "Management Fee").
- 9.3 Unless otherwise agreed, NHS Supply Chain will submit a quarterly reconciliation report detailing all Management Fee breakdowns. The Supplier shall then confirm if they agree with the eligible Management Fee charges. On receipt of the agreed Management Fee values, NHS Supply Chain will create a Management Fee Invoice.
- 9.4 NHS Supply Chain contacts the Supplier for valid copies of Trust Purchase Orders and Supplier Quotations for the relevant URN number.
- 9.5 NHS Supply Chain reconciliation will contain:
 - 9.5.1 Order Reference
 - 9.5.2 Trust Name
 - 9.5.3 Actual Order Date
 - 9.5.4 Customer PO Number
 - 9.5.5 Supplier Quotation Number
 - 9.5.6 Order Value (exc VAT) and Order Value minus shipping and handling
 - 9.5.7 Management Fee % Inc VAT
- 9.6 On receipt of this request the supplier will agree the values of the Management Fee reconciliation, and NHS Supply Chain shall invoice the Supplier for the Management Fee. The Supplier shall pay the Management Fee within thirty (30) days from receipt of such invoice.
- 9.7 Where the Supplier raises a query with respect to an invoice for the Management Fee, or NHS Supply Chain raises a query with respect to the Management Fee Report, the Supplier and NHS Supply Chain shall liaise with each other and agree a resolution to such query within thirty (30) days of the query being raised. If the Parties are unable to agree a resolution within thirty (30) days the Parties shall refer to dispute resolution in accordance with Clause 23 of this Schedule 2.

Other Payments

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 14 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

9.8 Where any payments are to be made under this Framework Agreement by either Party in addition to any payments to be made by Participating Authorities under any Contracts and the Management Fee to be paid by the Supplier, the details of such payments and the invoicing arrangements shall be set out in the Commercial Schedule.

10 Warranties

- 10.1 The Supplier warrants and undertakes that:
 - 10.1.1 it will comply with the terms of all Contracts entered into by Participating Authorities under this Framework Agreement;
 - 10.1.2 it will comply with the KPIs set out in Schedule 8;
 - 10.1.3 it will promptly respond to all requests for information regarding the Framework Agreement, the Goods and/or Services and any Contracts at the frequency and in the format that NHS Supply Chain may reasonably require;
 - 10.1.4 all information included within the Supplier's response to any documents issued by the Authority as part of the procurement relating to the award of this Framework Agreement (to include, without limitation, as referred to in the Specification in the Tender Response Document and Commercial Schedule) and all accompanying materials is accurate;
 - 10.1.5 it has the right and authority to enter into this Framework Agreement and that it has the capability and capacity to fulfil its obligations under this Framework Agreement;
 - 10.1.6 it is a properly constituted entity and it is fully empowered by the terms of its constitutional documents to enter into and to carry out its obligations under this Framework Agreement and the documents referred to in this Framework Agreement;
 - 10.1.7 all necessary actions to authorise the execution of and performance of its obligations under this Framework Agreement have been taken before such execution;
 - 10.1.8 there are no pending or threatened actions or proceedings before any court or administrative agency which would materially adversely affect the financial condition, business or operations of the Supplier;
 - 10.1.9 there are no material agreements existing to which the Supplier is a party which prevent the Supplier from entering into or complying with this Framework Agreement;
 - 10.1.10 it has and will continue to have the capacity, funding and cash flow to meet all its obligations under this Framework Agreement; and
 - 10.1.11 it has satisfied itself as to the nature and extent of the risks assumed by it under the Framework Agreement and has gathered all information necessary to perform its obligations under the Framework Agreement and all other obligations assumed by it.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 15 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 10.1.12 it shall: (i) comply with all relevant Law and Guidance and shall use Good Industry Practice to ensure that there is no slavery or human trafficking in its supply chains; and (ii) notify the Authority immediately if it becomes aware of any actual or suspected incidents of slavery or human trafficking in its supply chains;
- 10.1.13 it shall at all times conduct its business in a manner that is consistent with any anti-slavery Policy of the Authority and shall provide to the Authority any reports or other information that the Authority may request as evidence of the Supplier's compliance with this Clause 10.1.13 and/or as may be requested or otherwise required by the Authority in accordance with its antislavery Policy.
- 10.2 The Supplier warrants that all information, data and other records and documents required by NHS Supply Chain as set out in the Specification and Tender Response Document shall be submitted to NHS Supply Chain in the format and in accordance with any timescales set out in the Specification and Tender Response Document.
- 10.3 Unless the parties agree otherwise in writing, the Supplier warrants and undertakes to NHS Supply Chain that it shall comply with any E-Procurement Guidance as it may apply to the Supplier and shall carry out all reasonable acts required of the Supplier to enable NHS Supply Chain to comply with such E-Procurement Guidance.
- 10.4 The Supplier warrants and undertakes that at the Commencement Date it is not and throughout the term of the Framework Agreement and any Contracts it will not be, involved in any Occasion of Tax Non-compliance.
- 10.5 The Supplier further warrants and undertakes to NHS Supply Chain that it will inform NHS Supply Chain in writing immediately upon becoming aware that any of the warranties set out in Clause 10 of this Schedule 2 have been breached or there is a risk that any warranties may be breached.
- 10.6 Any warranties provided under this Framework Agreement are both independent and cumulative and may be enforced independently or collectively at the sole discretion of the enforcing Party.

11 Intellectual Property

11.1 Unless otherwise agreed in writing between the Parties, the Supplier has no right to use the branding or logo(s) of NHS Supply Chain or NHS in the promotion or marketing of the Supplier's goods and services, nor to reference the approval, support, endorsement, authorisation, certification or similar of NHS Supply Chain or NHS in relation to the Supplier's goods and services.

12 Statutory compliance

- 12.1 The Supplier shall comply with all Law and Guidance relevant to its obligations under this Framework Agreement and any Contracts.
- 12.2 Without limitation to Clause 12.1 of this Schedule 2, the Supplier shall be responsible for obtaining any statutory licences, authorisations, consents or permits required in connection with its performance of its obligations under this Framework Agreement and any Contracts.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 16 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

13 Independence of Participating Authorities

- 13.1 The Supplier acknowledges that each Participating Authority is independently responsible for the conduct of its award of Contracts under this Framework Agreement and that NHS Supply Chain is not responsible or accountable for and shall have no liability whatsoever in relation to:
 - 13.1.1 the conduct of Participating Authorities other than NHS Supply Chain in relation to the operation of this Framework Agreement; or
 - 13.1.2 the performance or non-performance of any Participating Authorities other than NHS Supply Chain under any Contracts between the Supplier and such other Participating Authorities entered into under this Framework Agreement.

14 Limitation of liability

- 14.1 Nothing in this Framework Agreement shall exclude or restrict the liability of either Party:
 - 14.1.1 for death or personal injury resulting from its negligence;
 - 14.1.2 for fraud or fraudulent misrepresentation;
 - 14.1.3 in any other circumstances where liability may not be limited or excluded under any applicable law; or
 - 14.1.4 to make any payments agreed in accordance with Clause 9 of this Schedule 2.
- 14.2 Subject to Clauses 14.1, 14.3 and 14.5 of this Schedule 2, the total liability of each Party to the other under or in connection with this Framework Agreement whether arising in contract, tort, negligence, breach of statutory duty or otherwise shall be limited in aggregate to five hundred thousand pounds (£500,000).
- 14.3 There shall be no right to claim losses, damages and/or other costs and expenses under or in connection with this Framework Agreement whether arising in contract (to include, without limitation, under any relevant indemnity), tort, negligence, breach of statutory duty or otherwise to the extent that any losses, damages and/or other costs and expenses claimed are in respect of loss of production, loss of business opportunity or are in respect of indirect loss of any nature suffered or alleged.
- 14.4 Each Party shall at all times take all reasonable steps to minimise and mitigate any loss for which one Party is entitled to bring a claim against the other pursuant to this Framework Agreement.
- 14.5 The liability of the Supplier and any Participating Authorities under any Contracts entered into pursuant to this Framework Agreement shall be as set out in the Call-off Terms and Conditions for the Supply of Goods forming part of such Contracts.

15 Insurance

15.1 Subject to Clauses 15.2 and 15.3 of this Schedule 2 and unless otherwise confirmed in writing by NHS Supply Chain, as a minimum level of protection, the Supplier shall

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 17 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

put in place and/or maintain in force at its own cost with a reputable commercial insurer, insurance arrangements in respect of employer's liability, public liability and product liability in accordance with Good Industry Practice with the minimum cover per claim being the greater of five million pounds (£5,000,000) or any sum as required by Law unless otherwise agreed with NHS Supply Chain in writing.

- 15.2 Without limitation to any insurance arrangements as required by Law, the Supplier shall put in place and/or maintain the different types and/or levels of indemnity arrangements explicitly required by NHS Supply Chain, if specified in the Key Provisions.
- 15.3 Provided that the Supplier maintains all indemnity arrangements required by Law, the Supplier may self insure in order to meet other relevant requirements referred to at Clauses 15.1 and 15.2 of this Schedule 2 on condition that such self insurance arrangements offer the appropriate levels of protection and are approved by NHS Supply Chain in writing prior to the Commencement Date.
- 15.4 The amount of any indemnity cover and/or self insurance arrangements shall not relieve the Supplier of any liabilities under this Framework Agreement. It shall be the responsibility of the Supplier to determine the amount of indemnity and/or self insurance cover that will be adequate to enable it to satisfy its potential liabilities under this Framework Agreement. Accordingly, the Supplier shall be liable to make good any deficiency if the proceeds of any indemnity cover and/or self insurance arrangement is insufficient to cover the settlement of any claim.
- 15.5 The Supplier warrants that it shall not take any action or fail to take any reasonable action or (in so far as it is reasonable and within its power) permit or allow others to take or fail to take any action, as a result of which its insurance cover may be rendered void, voidable, unenforceable, or be suspended or impaired in whole or in part, or which may otherwise render any sum paid out under such insurances repayable in whole or in part.
- 15.6 The Supplier shall from time to time and in any event within five (5) Business Days of written demand provide documentary evidence to NHS Supply Chain that insurance arrangements taken out by the Supplier pursuant to Clause 15 of this Schedule 2 and the Key Provisions are fully maintained and that any premiums on them and/or contributions in respect of them (if any) are fully paid.
- 15.7 Upon the expiry or earlier termination of this Framework Agreement, the Supplier shall ensure that any ongoing liability it has or may have arising out of this Framework Agreement shall continue to be the subject of appropriate indemnity arrangements for the period of twenty one (21) years from termination or expiry of this Framework Agreement or until such earlier date as that liability may reasonably be considered to have ceased to exist.

16 <u>Term and termination</u>

- 16.1 This Framework Agreement shall commence on the Commencement Date and, unless terminated earlier in accordance with the terms of this Framework Agreement or the general law, shall continue until the end of the Term.
- 16.2 NHS Supply Chain shall be entitled to extend the Term on one or more occasions by giving the Supplier written notice no less than three (3) months prior to the date on which this Framework Agreement would otherwise have expired, provided that the

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of			nt for the Supply of Goods
	Revision: 7	August 2017	Page 18 of 90
	NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

duration of this Framework Agreement shall be no longer than the total term specified in the Key Provisions.

- 16.3 In the case of a breach of any of the terms of this Framework Agreement by either Party that is capable of remedy (including any failure to pay sums due under this Framework Agreement), the non-breaching Party shall, without prejudice to its other rights and remedies under this Framework Agreement, issue notice of the breach and allow the Party in breach the opportunity to remedy such breach in the first instance via a remedial proposal put forward by the Party in breach ("**Remedial Proposal**") before exercising any right to terminate this Framework Agreement in accordance with Clause 16.4.1(ii) of this Schedule 2. Such Remedial Proposal must be agreed with the non-breaching Party (such agreement not to be unreasonably withheld or delayed) and must be implemented by the Party in breach in accordance with the timescales referred to in the agreed Remedial Proposal. Once agreed, any changes to a Remedial Proposal must be approved by the Parties in writing. Any failure by the Party in breach to:
 - 16.3.1 put forward and agree a Remedial Proposal with the non-breaching Party in relation to the relevant default or breach within a period of ten (10) Business Days (or such other period as the non-breaching Party may agree in writing) from written notification of the relevant default or breach from the non-breaching Party;
 - 16.3.2 comply with such Remedial Proposal (including, without limitation, as to its timescales for implementation, which shall be thirty (30) days unless otherwise agreed between the Parties); and/or
 - 16.3.3 remedy the default or breach notwithstanding the implementation of such Remedial Proposal in accordance with the agreed timescales for implementation,

shall be deemed, for the purposes of Clause 16.4.1(ii) of this Schedule 2, a material breach of this Framework Agreement by the Party in breach not remedied in accordance with an agreed Remedial Proposal.

- 16.4 Either Party may terminate this Framework Agreement forthwith by notice in writing to the other Party if such other Party:
 - 16.4.1 commits a material breach of any of the terms of this Framework Agreement which is:
 - (i) not capable of remedy; or
 - (ii) in the case of a breach capable of remedy, which is not remedied in accordance with a Remedial Proposal; or
 - 16.4.2 has been served with at least two (2) previous breach notices as a result of any material breaches which are capable of remedy within any twelve (12) month rolling period whether or not the Party in breach has remedied the breach in accordance with a Remedial Proposal. The twelve (12) months rolling period is the twelve (12) months immediately preceding the date of the third breach notice.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods Revision: 7 August 2017 Page 19 of 90		

- 16.5 NHS Supply Chain may terminate this Framework Agreement forthwith by notice in writing to the Supplier if:
 - 16.5.1 the Supplier, or any third party guaranteeing the obligations of the Supplier under this Framework Agreement, ceases or threatens to cease carrying on its business; suspends making payments on any of its debts or announces an intention to do so; is, or is deemed for the purposes of any Law to be, unable to pay its debts as they fall due or insolvent: enters into or proposes any composition, assignment or arrangement with its creditors generally; takes any step or suffers any step to be taken in relation to its winding-up, dissolution, administration (whether out of court or otherwise) or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise) otherwise than as part of, and exclusively for the purpose of, a bona fide reconstruction or amalgamation; has a liquidator, trustee in bankruptcy, judicial custodian, compulsory manager, receiver, administrative receiver, administrator or similar officer appointed (in each case, whether out of court or otherwise) in respect of it or any of its assets; has any security over any of its assets enforced; or any analogous procedure or step is taken in any jurisdiction;
 - 16.5.2 the Supplier undergoes a change of control within the meaning of sections 450 and 451 of the Corporation Tax Act 2010 (other than for an intra-group change of control) without the prior written consent of NHS Supply Chain and NHS Supply Chain shall be entitled to withhold such consent if, in the reasonable opinion of NHS Supply Chain, the proposed change of control will have a material impact on the performance of this Framework Agreement or the reputation of NHS Supply Chain;
 - 16.5.3 the Supplier purports to assign, Sub-contract, novate, create a trust in or otherwise transfer or dispose of this Framework Agreement in breach of Clause 29 of this Schedule 2;
 - 16.5.4 pursuant to and in accordance with the Key Provisions and Clauses 16.6, 24.8; 26.2; 26.4 and 30.2 of this Schedule 2; or
 - 16.5.5 the Supplier is in breach of Clause 10.4 of this Schedule 2.
- 16.6 If NHS Supply Chain, acting reasonably, has good cause to believe that there has been a material deterioration in the financial circumstances of the Supplier and/or any third party guaranteeing the obligations of the Supplier under this Framework Agreement and/or any material Sub-contractor of the Supplier when compared to any information provided to and/or assessed by NHS Supply Chain as part of any procurement process or other due diligence leading to the award of this Framework Agreement to the Supplier or the entering into a Sub-contract by the Supplier, the following process shall apply:
 - 16.6.1 NHS Supply Chain may (but shall not be obliged to) give notice to the Supplier requesting adequate financial or other security and/or assurances for due performance of its material obligations under this Framework Agreement on such reasonable and proportionate terms as NHS Supply Chain may require within a reasonable time period as specified in such notice;

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 20		Page 20 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 16.6.2 a failure or refusal by the Supplier to provide the financial or other security and/or assurances requested in accordance with Clause 16.6 of this Schedule 2 in accordance with any reasonable timescales specified in any such notice issued by NHS Supply Chain shall be deemed a breach of this Framework Agreement by the Supplier and shall be referred to and resolved in accordance with the Dispute Resolution Procedure; and
- 16.6.3 a failure to resolve such breach in accordance with such Dispute Resolution Procedure by the end of the escalation stage of such process (as set out in Clause 23.2 of this Schedule 2) shall entitle, but shall not compel, NHS Supply Chain to terminate this Framework Agreement in accordance with Clause 16.4.1(i) of this Schedule 2.
- 16.6.4 In order that NHS Supply Chain may act reasonably in exercising its discretion in accordance with Clause 16.6 of this Schedule 2, the Supplier shall provide NHS Supply Chain with such reasonable and proportionate up-to-date financial or other information relating to the Supplier or any relevant third party entity upon request.
- 16.7 NHS Supply Chain may terminate this Framework Agreement forthwith by notice in writing to the Supplier where:
 - 16.7.1 the Framework Agreement has been substantially amended to the extent that the Public Contracts Regulations 2015 require a new procurement procedure;
 - 16.7.2 the Authority has become aware that the Supplier should have been excluded under Regulation 57(1) or (2) of the Public Contracts Regulations 2015 from the procurement procedure leading to the award of this Framework Agreement;
 - 16.7.3 the Framework Agreement should not have been awarded to the Supplier in view of a serious infringement of obligations under European law declared by the Court of Justice of the European Union under Article 258 of the Treaty on the Functioning of the EU; or
 - 16.7.4 there has been a failure by the Supplier and/or one its Sub-contractors to comply with legal obligations in the fields of environmental, social or labour Law. Where the failure to comply with legal obligations in the fields of environmental, social or labour Law is a failure by one of the Supplier's Sub-contractors, the Authority may request the replacement of such Sub-contractor and the Supplier shall comply with such request as an alternative to the Authority terminating this Framework Agreement under this Clause 16.7.4.

17 Consequences of expiry or earlier termination of this Framework Agreement

17.1 Upon expiry or earlier termination of this Framework Agreement, NHS Supply Chain and the Supplier agree that all Contracts entered into under this Framework Agreement

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017		Page 21 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

will continue in full force and effect unless otherwise terminated under the terms and conditions of such Contracts.

- 17.2 The Supplier agrees that where this Framework Agreement has been terminated properly in accordance with Clause 16 of this Schedule 2 it shall not be entitled to make a claim against NHS Supply Chain in relation to costs incurred in the provision of the Goods and/or Services which do not form part of the Contract Price paid or payable by an Authority.
- 17.3 The Supplier shall cooperate fully with NHS Supply Chain or, as the case may be, any replacement supplier during any re-procurement and handover period prior to and following the expiry or earlier termination of this Framework Agreement. This cooperation shall extend to providing access to all information relevant to the operation of this Framework Agreement, as reasonably required by NHS Supply Chain to achieve a fair and transparent re-procurement and/or an effective transition without disruption to routine operational requirements.
- 17.4 The expiry or earlier termination of this Framework Agreement for whatever reason shall not affect any rights or obligations of either Party which accrued prior to such expiry or earlier termination.
- 17.5 The expiry or earlier termination of this Framework Agreement shall not affect any obligations which expressly or by implication are intended to come into or continue in force on or after such expiry or earlier termination.

18 Suspension of Supplier's appointment

- 18.1 Without prejudice to NHS Supply Chain's rights to terminate this Framework Agreement, if a right for NHS Supply Chain to terminate this Framework Agreement arises (irrespective of whether the circumstances leading to such right are capable of remedy) in accordance with Clause 16 of this Schedule 2, NHS Supply Chain may suspend the Supplier's appointment to receive new Orders under this Framework Agreement by giving notice in writing to the Supplier and all Participating Authorities.
- 18.2 If NHS Supply Chain provides notice to the Supplier in accordance with Clause 18.1 of this Schedule 2, the Supplier's appointment shall be suspended for the period set out in the notice or such other period notified to the Supplier by NHS Supply Chain in writing from time to time provided that such suspension shall be lifted if:
 - 18.2.1 the circumstances leading to NHS Supply Chain's right to terminate this Framework Agreement have been remedied;
 - 18.2.2 NHS Supply Chain has satisfied itself that the risk and/or impact of the circumstances giving rise to NHS Supply Chain's right to terminate this Framework Agreement no longer requires such suspension; or
 - 18.2.3 NHS Supply Chain exercises its rights to terminate this Framework Agreement in accordance with Clause 16 of this Schedule 2.

19 <u>Complaints process</u>

19.1 The Supplier shall notify NHS Supply Chain of any formal written complaints made by other Participating Authorities relating to the Supplier's noncompliance with any of its

LEGAL TEMP 810	ent for the Supply of Goods	
Revision: 7	August 2017	Page 22 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

obligations under any Contract within two (2) Business Days of the Supplier becoming aware of such complaints.

- 19.2 Without prejudice to any rights and remedies that the Participating Authority may have under the relevant Contract and/or NHS Supply Chain may have under this Framework Agreement, the Supplier shall use its reasonable endeavours to resolve such complaint within ten (10) Business Days and in so doing, shall deal with the complaint fully, expeditiously and fairly.
- 19.3 Within two (2) Business Days of a written request by NHS Supply Chain, the Supplier shall provide further reasonable details of the complaint to NHS Supply Chain, including details of the steps being taken to progress its resolution and, following its resolution, details of how and when the complaint was resolved.

20 Sustainable development

- 20.1 The Supplier shall comply in all material respects with applicable environmental and social Law requirements in force from time to time in relation to the Goods and Services. Where the provisions of any such Law are implemented by the use of voluntary agreements, the Supplier shall comply with such agreements as if they were incorporated into English law subject to those voluntary agreements being cited in the Specification. Without prejudice to the generality of the foregoing, the Supplier shall:
 - 20.1.1 comply with all Policies and/or procedures and requirements set out in the Specification in relation to any stated environmental and social requirements, characteristics and impacts of the Goods and Services and the Supplier's supply chain;
 - 20.1.2 maintain relevant policy statements documenting the Supplier's significant social and environmental aspects as relevant to the Goods and Services being supplied and as proportionate to the nature and scale of the Supplier's business operations; and
 - 20.1.3 maintain plans and procedures that support the commitments made as part of the Supplier's significant social and environmental policies, as referred to in Clause 20.1.2 of this Schedule 2.
- 20.2 The Supplier shall meet reasonable requests by NHS Supply Chain for information evidencing the Supplier's compliance with the provisions of Clause 20 of this Schedule 2.

21 Electronic product information

- 21.1 Where requested by NHS Supply Chain, the Supplier shall provide NHS Supply Chain with the Product Information in such manner and upon such media as agreed between the Supplier and NHS Supply Chain from time to time for the sole use by NHS Supply Chain.
- 21.2 The Supplier warrants that the Product Information is complete and accurate as at the date upon which it is delivered to NHS Supply Chain and that the Product Information shall not contain any data or statement which gives rise to any liability on the part of NHS Supply Chain following publication of the same in accordance with Clause 21 of this Schedule 2.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 23 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 21.3 If the Product Information ceases to be complete and accurate, the Supplier shall promptly notify NHS Supply Chain in writing of any modification or addition to or any inaccuracy or omission in the Product Information.
- 21.4 The Supplier grants NHS Supply Chain a perpetual, non-exclusive, royalty free licence to use and exploit the Product Information and any Intellectual Property Rights in the Product Information for the purpose of illustrating the range of goods and services (including, without limitation, the Goods and Services) available in NHS Supply Chain's product catalogue in relation to any catalogues produced during the Term. Subject to Clause 21.5 of this Schedule 2, no right to illustrate or advertise the Product Information is granted to the Supplier by the Authority, as a consequence of the licence conferred by this Clause 21.4 of this Schedule 2.
- 21.5 NHS Supply Chain may reproduce for its sole use the Product Information provided by the Supplier in NHS Supply Chain's product catalogue from time to time which may be made available on any healthcare communications networks in electronic format and/or made available on NHS Supply Chain's external website and/or made available on other digital media from time to time.
- 21.6 For the avoidance of doubt the Supplier shall have no right to compel NHS Supply Chain to exhibit the Product Information in any product catalogue as a result of the approval given by it pursuant to this Clause 21.6 of this Schedule 2 or otherwise under the terms of this Framework Agreement.
- 21.7 NHS Supply Chain may approach the Supplier during the Term to offer the Supplier the opportunity to take part in specific promotions or to purchase additional advertising space in relation to the Goods and/or Services, the Framework Agreement and any Contract and the Parties shall agree an appropriate price for any such advertising. If any such opportunity is cancelled by NHS Supply Chain it shall refund the purchase price to the Supplier but for the avoidance of doubt, NHS Supply Chain shall not be liable for any incidental costs incurred by the Supplier, including costs associated with the development of an advert.
- 21.8 The Supplier agrees to indemnify and keep indemnified NHS Supply Chain against any loss, damages, costs, expenses (including without limitation legal costs and expenses), claims or proceedings arising out of or in connection with NHS Supply Chain's use of the Product Information, provided always that NHS Supply Chain has not materially misused the Product Information.

22 Change management

22.1 The Supplier acknowledges to NHS Supply Chain that the requirements for the Goods and Services may change during the Term and the Supplier shall not unreasonably withhold or delay its consent to any reasonable variation or addition to the Specification and Tender Response Document, as may be requested by NHS Supply Chain from time to time. Any change to the Goods and Services or other variation to this Framework Agreement shall only be binding once it has been agreed in writing and signed by an authorised representative of both Parties.

23 Dispute resolution

23.1 During any dispute, including a dispute as to the validity of this Framework Agreement, it is agreed that the Supplier shall continue its performance of the provisions of the Framework Agreement to the extent that such obligations are not the subject of the

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 24 of		Page 24 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

dispute (unless NHS Supply Chain requests in writing that the Supplier does not do so).

- 23.2 In the case of a dispute arising out of or in connection with this Framework Agreement the Supplier and NHS Supply Chain shall make every reasonable effort to communicate and cooperate with each other with a view to resolving the dispute and follow the procedure set out in Clause 23.2 of this Schedule 2 as the first stage in the Dispute Resolution Procedure.
- 23.3 If any dispute arises out of the Framework Agreement either Party may serve a notice on the other Party to commence formal resolution of the dispute. The Parties shall first seek to resolve the dispute by escalation in accordance with the management levels as set out in Clause 5 of the Key Provisions. Respective representatives, at each level as set out in Clause 5 of the Key Provisions, shall have five (5) Business Days at each level during which they will use their reasonable endeavours to resolve the dispute before escalating the matter to the next level as appropriate until all levels have been exhausted. Level 1 will commence on the date of service of the dispute notice. The final level of the escalation process shall be deemed exhausted on the expiry of five (5) Business Days following escalation to that level unless otherwise agreed by the Parties in writing.
- 23.4 If the procedure set out in Clause 23.2 of this Schedule 2 above has been exhausted and fails to resolve such dispute, as part of the Dispute Resolution Procedure, the Parties will attempt to settle it by mediation. The Parties shall, acting reasonably, attempt to agree upon a mediator. In the event that the Parties fail to agree a mediator within five (5) Business Days following the exhaustion of all levels of the escalation procedure at Clause 22.3 of this Schedule 2, the mediator shall be nominated and confirmed by the Centre for Effective Dispute Resolution, London.
- 23.5 The mediation shall commence within twenty eight (28) days of the conformation of the mediator in accordance with Clause 22.4 of this Schedule 2 or at such other time as may be agreed by the Parties in writing. Neither Party will terminate such mediation process until each Party has made its opening presentation and the mediator has met each Party separately for at least one hour or one Party has failed to participate in the mediation process. After this time, either Party may terminate the mediation process by notice to the other Party (such notification may be verbal provided that it is followed up by written conformation). NHS Supply Chain and the Supplier will cooperate with any person appointed as mediator providing them with such information and other assistance as they shall require and will pay their costs, as they shall determine or in the absence of such determination such costs will be shared equally.
- 23.6 Nothing in this Framework Agreement shall prevent:
 - 23.6.1 NHS Supply Chain taking action in any court in relation to any death or personal injury arising or allegedly arising in connection with supply of the Goods and/or Services; or
 - 23.6.2 either Party seeking from any court any interim or provisional relief that may be necessary to protect the rights or property of that Party or that relates to the safety of patients or the security of Confidential Information, pending resolution of the relevant dispute in accordance with the Dispute Resolution Procedure.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 25 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

23.7 Clause 23 of this Schedule 2 shall survive the expiry of or earlier termination of this Framework Agreement for any reason.

24 Force majeure

- 24.1 Subject to Clause 24.2 of this Schedule 2 neither Party shall be liable to the other for any failure to perform all or any of its obligations under this Framework Agreement nor liable to the other Party for any loss or damage arising out of the failure to perform its obligations to the extent only that such performance is rendered impossible by a Force Majeure Event.
- 24.2 The Supplier shall only be entitled to rely on a Force Majeure Event and the relief set out in Clause 24 of this Schedule 2 and will not be considered to be in default or liable for breach of any obligations under this Framework Agreement if:
 - 24.2.1 the Supplier has fulfilled its obligations pursuant to Clause 6 of this Schedule 2;
 - 24.2.2 the Force Majeure Event does not arise directly or indirectly as a result of any wilful or negligent act or default of the Supplier; and
 - 24.2.3 the Supplier has complied with the procedural requirements set out in Clause 24 of this Schedule 2.
- 24.3 Where a Party is (or claims to be) affected by a Force Majeure Event it shall use reasonable endeavours to mitigate the consequences of such a Force Majeure Event upon the performance of its obligations under this Framework Agreement and to resume the performance of its obligations affected by the Force Majeure Event as soon as practicable.
- 24.4 Where the Force Majeure Event affects the Supplier's ability to perform part of its obligations under the Framework Agreement the Supplier shall fulfil all such contractual obligations that are not so affected and shall not be relieved from its liability to do so.
- 24.5 If either Party is prevented or delayed in the performance of its obligations under this Framework Agreement by a Force Majeure Event, that Party shall as soon as reasonably practicable serve notice in writing on the other Party specifying the nature and extent of the circumstances giving rise to its failure to perform or any anticipated delay in performance of its obligations.
- 24.6 Subject to service of such notice, the Party affected by such circumstances shall have no liability for its failure to perform or for any delay in performance of its obligations affected by the Force Majeure Event only for so long as such circumstances continue and for such time after they cease as is necessary for that Party, using its best endeavours, to recommence its affected operations in order for it to perform its obligations.
- 24.7 The Party claiming relief shall notify the other in writing as soon as the consequences of the Force Majeure Event have ceased and of when performance of its affected obligations can be resumed.
- 24.8 If the Supplier is prevented from performance of its obligations as a result of a Force Majeure Event, NHS Supply Chain may at any time if the Force Majeure Event subsists

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 26 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

for thirty (30) days or more, terminate this Framework Agreement on service of written notice on the Supplier.

- 24.9 Following such termination in accordance with Clause 24.8 of this Schedule 2 and subject to Clause 24.10 of this Schedule 2, neither Party shall have any liability to the other.
- 24.10 Any rights and liabilities of either Party which accrued prior to such termination in accordance with Clause 24.8 of this Schedule 2 shall continue in full force and effect unless otherwise specified in this Framework Agreement.

25 Records retention and right of audit

- 25.1 Subject to any statutory requirement and Clause 25.2 of this Schedule 2, the Supplier shall keep secure and maintain for the Term and six (6) years afterwards, or such longer period as may be agreed between the Parties, full and accurate records of all matters relating to this Framework Agreement.
- 25.2 Where any records could be relevant to a claim for personal injury such records shall be kept secure and maintained for a period of twenty one (21) years from the date of expiry or earlier termination of this Framework Agreement.
- 25.3 NHS Supply Chain shall have the right to audit the Supplier's compliance with this Framework Agreement. The Supplier shall permit or procure permission for NHS Supply Chain or its authorised representative during normal business hours having given advance written notice of no less than five (5) Business Days, access to any premises and facilities, books and records reasonably required to audit the Supplier's compliance with its obligations under this Framework Agreement.
- 25.4 Should the Supplier Sub-contract any of its obligations under this Framework Agreement, NHS Supply Chain shall have the right to audit and inspect such third party. The Supplier shall procure permission for NHS Supply Chain or its authorised representative during normal business hours no more than once in any twelve (12) months, having given advance written notice of no less than five (5) Business Days, access to any premises and facilities, books and records used in the performance of the Supplier's obligations under this Framework Agreement that are Sub-contracted to such third party. The Supplier shall cooperate with such audit and inspection and accompany NHS Supply Chain or its authorised representative if requested.
- 25.5 The Supplier shall grant to NHS Supply Chain or its authorised representative, such access to those records as they may reasonably require in order to check the Supplier's compliance with this Framework Agreement for the purposes of:
 - 25.5.1 the examination and certification of NHS Supply Chain's accounts; or
 - 25.5.2 any examination pursuant to section 6(1) of the National Audit Act 1983 of the economic efficiency and effectiveness with which NHS Supply Chain has used its resources.
- 25.6 The Comptroller and Auditor General may examine such documents as they may reasonably require which are owned, held or otherwise within the control of the Supplier and may require the Supplier to provide such oral and/or written explanations as they consider necessary. Clause 25 of this Schedule 2 does not constitute a

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goo		
Revision: 7	August 2017	Page 27 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

requirement or agreement for the examination, certification or inspection of the accounts of the Supplier under section 6(3)(d) and 6(5) of the National Audit Act 1983.

- 25.7 The Supplier shall provide reasonable cooperation to NHS Supply Chain, its representatives and any regulatory body in relation to any audit, review, investigation or enquiry carried out in relation to the subject matter of this Framework Agreement.
- 25.8 The Supplier shall provide all reasonable information as may be reasonably requested by NHS Supply Chain to evidence the Supplier's compliance with the requirements of this Framework Agreement.

26 Conflicts of interest and the prevention of fraud

- 26.1 The Supplier shall take appropriate steps to ensure that neither the Supplier nor any Staff are placed in a position where, in the reasonable opinion of NHS Supply Chain, there is or may be an actual conflict, or a potential conflict, between the pecuniary or personal interests of the Supplier and the duties owed to NHS Supply Chain under the provisions of this Framework Agreement. The Supplier will disclose to NHS Supply Chain full particulars of any such conflict of interest which may arise.
- 26.2 NHS Supply Chain reserves the right to terminate this Framework Agreement immediately by notice in writing and/or to take such other steps it deems necessary where, in the reasonable opinion of NHS Supply Chain, there is or may be an actual conflict, or a potential conflict, between the pecuniary or personal interests of the Supplier and the duties owed to NHS Supply Chain under the provisions of this Framework Agreement. The actions of NHS Supply Chain pursuant to this Clause 26.2 of this Schedule 2 shall not prejudice or affect any right of action or remedy which shall have accrued or shall subsequently accrue to NHS Supply Chain.
- 26.3 The Supplier shall take all reasonable steps to prevent Fraud by Staff and the Supplier (including its owners, members and directors). The Supplier shall notify NHS Supply Chain immediately if it has reason to suspect that any Fraud has occurred or is occurring or is likely to occur.
- 26.4 If the Supplier or its Staff commits Fraud NHS Supply Chain may terminate this Framework Agreement and recover from the Supplier the amount of any direct loss suffered by NHS Supply Chain resulting from the termination.

27 Equality and human rights

- 27.1 The Supplier shall:
 - 27.1.1 ensure that (a) it does not, whether as employer or as supplier of the Goods and Services, engage in any act or omission that would contravene the Equality Legislation, and (b) it complies with all its obligations as an employer or supplier of the Goods and Services as set out in the Equality Legislation and take reasonable endeavours to ensure its Staff do not unlawfully discriminate within the meaning of the Equality Legislation;
 - 27.1.2 in the management of its affairs and the development of its equality and diversity policies, cooperate with NHS Supply Chain in light of NHS Supply Chain's obligations to comply with its statutory equality duties whether under the Equality Act 2010 or otherwise. The Supplier shall take such reasonable and proportionate steps as NHS Supply Chain considers appropriate to

LEGAL TEMP 81	ent for the Supply of Goods	
Revision: 7	August 2017	Page 28 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

promote equality and diversity, including race equality, equality of opportunity for disabled people, gender equality, and equality relating to religion and belief, sexual orientation and age; and

- 27.1.3 the Supplier shall impose on all its Sub-contractors and suppliers, obligations substantially similar to those imposed on the Supplier by Clause 27 of this Schedule 2.
- 27.2 The Supplier shall meet reasonable requests by NHS Supply Chain for information evidencing the Supplier's compliance with the provisions of Clause 27 of this Schedule 2

28 Notice

- 28.1 Subject to Clause 23.5 of this Schedule 2, any notice required to be given by either Party under this Framework Agreement shall be in writing quoting the date of the Framework Agreement and shall be delivered by hand or sent by prepaid first class recorded delivery or by email to the person referred to in the Key Provisions or such other person as one Party may inform the other Party in writing from time to time.
- 28.2 A notice shall be treated as having been received:
 - 28.2.1 if delivered by hand within normal business hours when so delivered or, if delivered by hand outside normal business hours, at the next start of normal business hours; or
 - 28.2.2 if sent by first class recorded delivery mail on a normal Business Day, at 9.00 am on the second Business Day subsequent to the day of posting, or, if the notice was not posted on a Business Day, at 9.00 am on the third Business Day subsequent to the day of posting; or
 - 28.2.3 if sent by email, if sent within normal business hours when so sent or, if sent outside normal business hours, at the next start of normal business hours provided the sender has either received an electronic confirmation of delivery or has telephoned the recipient to inform the recipient that the email has been sent.

29 Assignment, novation and Sub-contracting

- 29.1 Subject to Clause 29.2 of this Schedule 2, the Supplier shall not assign, Sub-contract, novate, create a trust in, or in any other way dispose of the whole or any part of this Framework Agreement without the prior consent in writing of NHS Supply Chain, such consent not to be unreasonably withheld or delayed. If the Supplier Sub-contracts any of its obligations under this Framework Agreement, every act or omission of the Sub-contractor shall for the purposes of this Framework Agreement be deemed to be the act or omission of the Supplier and the Supplier shall be liable to NHS Supply Chain as if such act or omission had been committed or omitted by the Supplier itself.
- 29.2 The Supplier may assign, Sub-contract or novate this Framework Agreement to a member of its Group, provided always that such Group member shall have been assessed by NHS Supply Chain and passed to the satisfaction of NHS Supply Chain all grounds for exclusion and shortlisting criteria to be awarded onto this Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 29 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 29.3 Any authority given by NHS Supply Chain for the Supplier to Sub-contract any of its obligations under this Framework Agreement shall not impose any duty on NHS Supply Chain to enquire as to the competency of any authorised Sub-contractor. The Supplier shall ensure that any authorised Sub-contractor has the appropriate capability and capacity to perform the relevant obligations and that the obligations carried out by such Sub-contractor are fully in accordance with this Framework Agreement.
- 29.4 Where the Authority considers that the grounds for exclusion under Regulation 57 of the Public Contracts Regulations 2015 apply to any Sub-contractor, then:
 - 29.4.1 if the Authority finds there are compulsory grounds for exclusion, the Supplier shall ensure, or shall procure, that such Sub-contractor is replaced or not appointed; or
 - 29.4.2 if the Authority finds there are non-compulsory grounds for exclusion, the Authority may require the Supplier to ensure, or to procure, that such Subcontractor is replaced or not appointed and the Supplier shall comply with such a requirement.
- 29.5 NHS Supply Chain shall upon written request have the right to review any Sub-contract entered into by the Supplier in respect of the provision of the Goods and the Supplier shall provide a certified copy of any Sub-contract within five (5) Business Days of the date of a written request from NHS Supply Chain. For the avoidance of doubt, the Supplier shall have the right to redact any confidential pricing information in relation to such copies of Sub-contracts.
- 29.6 NHS Supply Chain may at any time transfer, assign, novate, Sub-contract or otherwise dispose of its rights and obligations under this Framework Agreement or any part of this Framework Agreement and the Supplier warrants that it will carry out all such reasonable further acts required to effect such transfer, assignment, novation, Sub-contracting or disposal. If NHS Supply Chain novates this Framework Agreement to any body that is not a Contracting Authority, from the effective date of such novation, the party assuming the position of NHS Supply Chain shall not further transfer, assign, novate, Sub-contract or otherwise dispose of its rights and obligations under this Framework Agreement or any part of this Framework Agreement without the prior written consent of the Supplier, such consent not to be unreasonably withheld or delayed by the Supplier.

30 Prohibited Acts

- 30.1 The Supplier warrants and represents that:
 - 30.1.1 it has not committed any offence under the Bribery Act 2010 or done any of the following ("**Prohibited Acts**"):
 - (i) offered, given or agreed to give any officer or employee of NHS Supply Chain any gift or consideration of any kind as an inducement or reward for doing or not doing or for having done or not having done any act in relation to the obtaining or performance of this or any other agreement with NHS Supply Chain or for showing or not

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 30 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

showing favour or disfavour to any person in relation to this or any other agreement with NHS Supply Chain; or

- (ii) in connection with this Framework Agreement paid or agreed to pay any commission other than a payment, particulars of which (including the terms and conditions of the agreement for its payment) have been disclosed in writing to NHS Supply Chain; and
- 30.1.2 it has in place adequate procedures to prevent bribery and corruption, as contemplated by section 7 of the Bribery Act 2010.
- 30.2 If the Supplier or its Staff (or anyone acting on its or their behalf) has done or does any of the Prohibited Acts or has committed or commits any offence under the Bribery Act 2010 with or without the knowledge of the Supplier in relation to this or any other agreement with NHS Supply Chain:
 - 30.2.1 NHS Supply Chain shall be entitled:
 - to terminate this Framework Agreement and recover from the Supplier the amount of any loss resulting from the termination;
 - to recover from the Supplier the amount or value of any gift, consideration or commission concerned; and
 - to recover from the Supplier any other loss or expense sustained in consequence of the carrying out of the Prohibited Act or the commission of the offence under the Bribery Act 2010;
 - 30.2.2 any termination under Clause 30.2.1 of this Schedule 2 shall be without prejudice to any right or remedy that has already accrued, or subsequently accrues, to NHS Supply Chain; and
 - 30.2.3 notwithstanding Clause 23 of this Schedule 2, any dispute relating to:
 - (i) the interpretation of Clause 30 of this Schedule 2; or
 - (ii) the amount or value of any gift, consideration or commission,

shall be determined by NHS Supply Chain, acting reasonably, and the decision shall be final and conclusive.

31 <u>General</u>

- 31.1 Each of the Parties is independent of the other and nothing contained in this Framework Agreement shall be construed to imply that there is any relationship between the Parties of partnership or of principal/agent or of employer/employee nor are the Parties hereby engaging in a joint venture and accordingly neither of the Parties shall have any right or authority to act on behalf of the other nor to bind the other by agreement or otherwise, unless expressly permitted by the terms of this Framework Agreement.
- 31.2 Failure or delay by either Party to exercise an option or right conferred by this Framework Agreement shall not of itself constitute a waiver of such option or right.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			nt for the Supply of Goods
Revision: 7		August 2017	Page 31 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

- 31.3 The delay or failure by either Party to insist upon the strict performance of any provision, term or condition of this Framework Agreement or to exercise any right or remedy consequent upon such breach shall not constitute a waiver of any such breach or any subsequent breach of such provision, term or condition.
- 31.4 Any provision of this Framework Agreement which is held to be invalid or unenforceable in any jurisdiction shall be ineffective to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining provisions of this Framework Agreement and any such invalidity or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provisions in any other jurisdiction.
- 31.5 Each Party acknowledges and agrees that it has not relied on any representation, warranty or undertaking (whether written or oral) in relation to the subject matter of this Framework Agreement and therefore irrevocably and unconditionally waives any rights it may have to claim damages against the other Party for any misrepresentation or undertaking (whether made carelessly or not) or for breach of any warranty unless the representation, undertaking or warranty relied upon is set out in this Framework Agreement or unless such representation, undertaking or warranty was made fraudulently.
- 31.6 Each Party shall bear its own expenses in relation to the preparation and execution of this Framework Agreement including all costs, legal fees and other expenses so incurred.
- 31.7 The rights and remedies provided in this Framework Agreement are cumulative and not exclusive of any rights or remedies provided by general law, or by any other contract or document. In this Clause 31.7 of this Schedule 2, right includes any power, privilege, remedy, or proprietary or security interest.
- 31.8 No persons other than the parties to this Framework Agreement and any Participating Authorities shall have the right to enforce the terms of this Framework Agreement which confer a benefit on such person or be entitled to object to or be required to consent to any amendment to the provisions of this Framework Agreement.
- 31.9 This Framework Agreement, any variation in writing signed by an authorised representative of each Party and any document referred to explicitly in this Framework Agreement or any variation to this Framework Agreement, contain the entire understanding between the Supplier and NHS Supply Chain relating to the operation of this Framework Agreement to the exclusion of all previous agreements, confirmations and understandings and there are no promises, terms, conditions or obligations whether oral or written, express or implied other than those contained or referred to in this Framework Agreement. Nothing in this Framework Agreement seeks to exclude either Party's liability for Fraud. Any tender conditions and/or disclaimers set out in the Authority's procurement documentation leading to the award of this Framework Agreement shall form part of this Framework Agreement.
- 31.10 This Framework Agreement, and any dispute or claim arising out of or in connection with it or its subject matter (including any non-contractual claims), shall be governed by, and construed in accordance with, the laws of England and Wales.
- 31.11 Subject to Clause 23 of this Schedule 2, the Parties irrevocably agree that the courts of England and Wales shall have non-exclusive jurisdiction to settle any dispute or

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Good		
Revision: 7	August 2017	Page 32 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

claim that arises out of or in connection with this Framework Agreement or its subject matter.

31.12 All written and oral communications and all written material referred to under this Framework Agreement shall be in English.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 33 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 3

Information Governance Provisions

1 Confidentiality

- 1.1 In respect of any Confidential Information it may receive directly or indirectly from the other Party ("Discloser") and subject always to the remainder of Clause 1 of this Schedule 3, each Party ("Recipient") undertakes to keep secret and strictly confidential and shall not disclose any such Confidential Information to any third party without the Discloser's prior written consent provided that:
 - 1.1.1 the Recipient shall not be prevented from using any general knowledge, experience or skills which were in its possession prior to the Commencement Date;
 - 1.1.2 the provisions of Clause 1 of this Schedule 3 shall not apply to any Confidential Information:
 - which is in or enters the public domain other than by breach of this Framework Agreement or other act or omissions of the Recipient;
 - which is obtained from a third party who is lawfully authorised to disclose such information without any obligation of confidentiality;
 - (iii) which is authorised for disclosure by the prior written consent of the Discloser;
 - (iv) which the Recipient can demonstrate was in its possession without any obligation of confidentiality prior to receipt of the Confidential Information from the Discloser; or
 - (v) which the Recipient is required to disclose purely to the extent to comply with the requirements of any relevant stock exchange.
- 1.2 Nothing in Clause 1 of this Schedule 3 shall prevent the Recipient from disclosing Confidential Information where it is required to do so by judicial, administrative, governmental or regulatory process in connection with any action, suit, proceedings or claim or otherwise by applicable Law, including the Freedom of Information Act 2000 ("FOIA"), Codes of Practice on Access to Government Information, on the Discharge of Public Authorities' Functions or on the Management of Records ("Codes of Practice") or the Environmental Information Regulations 2004 ("Environmental Regulations").
- 1.3 NHS Supply Chain may disclose the Supplier's Confidential Information:
 - 1.3.1 on a confidential basis to, any Contracting Authority (the Parties agree that all Contracting Authorities receiving such Confidential Information shall be entitled to further disclose the Confidential Information to other Contracting Authorities on the basis that the information is confidential and is not to be disclosed to a third party which is not part of any Contracting Authority);

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 34 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 1.3.2 on a confidential basis, to any consultant, contractor or other person engaged by NHS Supply Chain and/or the Contracting Authority receiving such information;
- 1.3.3 to any relevant party for the purpose of the examination and certification of NHS Supply Chain's accounts;
- 1.3.4 to any relevant party for any examination pursuant to section 6(1) of the National Audit Act 1983 of the economy, efficiency and effectiveness with which NHS Supply Chain has used its resources;
- 1.3.5 to Parliament and Parliamentary Committees or if required by any Parliamentary reporting requirements; or
- 1.3.6 on a confidential basis, to a proposed successor body in connection with any proposed or actual, assignment, novation or other disposal of rights, obligations, liabilities or property in connection with this Framework Agreement;

and for the purpose of this Framework Agreement, references to disclosure "on a confidential basis" shall mean NHS Supply Chain making clear the confidential nature of such information and that it must not be further disclosed except in accordance with Law or this Clause 1.3 of this Schedule 3.

- 1.4 The Supplier may only disclose NHS Supply Chain's Confidential Information, and any other information provided to the Supplier by NHS Supply Chain in relation to the operation of this Framework Agreement, to the Supplier's Staff or professional advisors who are directly involved in the performance of or advising on the Supplier's obligations under this Framework Agreement. The Supplier shall ensure that such Staff or professional advisors are aware of and shall comply with the obligations in Clause 1 of this Schedule 3 as to confidentiality and that all information, including Confidential Information, is held securely, protected against unauthorised use or loss and, at NHS Supply Chain's written discretion, destroyed securely or returned to NHS Supply Chain when it is no longer required. The Supplier shall not, and shall ensure that the Staff do not, use any of NHS Supply Chain's Confidential Information received otherwise than for the purposes of performing the Supplier's obligations in this Framework Agreement.
- 1.5 Nothing in this Clause 1 of this Schedule 3 shall prevent the Recipient from disclosing the Confidential Information to its Group companies, provided that the Recipient procures that such Group companies comply with this Clause 1 of this Schedule 3 as if each reference to the Recipient in this Clause 1 of this Schedule 3 is a reference to any such Group company receiving the Confidential Information.
- 1.6 For the avoidance of doubt, save as required by Law or as otherwise set out in this Schedule 3, the Supplier shall not, without the prior written consent of NHS Supply Chain (such consent not to be unreasonably withheld or delayed), announce that it has entered into this Framework Agreement and/or that it has been appointed as a Supplier to NHS Supply Chain and/or make any other announcements about this Framework Agreement.
- 1.7 Clause 1 of this Schedule 3 shall remain in force:

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 35 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 1.7.1 without limit in time in respect of Confidential Information which comprises Personal Data, Sensitive Personal Data or which relates to national security; and
- 1.7.2 for all other Confidential Information for a period of three (3) years after the expiry or earlier termination of this Framework Agreement unless otherwise agreed in writing by the Parties.

2 Data protection

- 2.1 The Parties acknowledge their respective duties under Data Protection Legislation and shall give each other all reasonable assistance as appropriate or necessary to enable each other to comply with those duties.
- 2.2 Where the Supplier is Processing Personal Data under or in connection with this Framework Agreement, the Supplier must, in particular, but without limitation:
 - 2.2.1 only Process such Personal Data as is necessary to perform its obligations under this Framework Agreement, and only in accordance with any instructions given by NHS Supply Chain under this Framework Agreement;
 - 2.2.2 put in place appropriate technical and organisational measures against any unauthorised or unlawful Processing of that Personal Data, and against the accidental loss or destruction of or damage to such Personal Data having regard to the specific requirements of Clause 2 of this Schedule 3, the state of technical development and the level of harm that may be suffered by a Data Subject whose Personal Data is affected by unauthorised or unlawful Processing or by its loss, damage or destruction;
 - 2.2.3 take reasonable steps to ensure the reliability of Staff who will have access to Personal Data, and ensure that those Staff are aware of and trained in the policies and procedures identified in Clause 2 of this Schedule 3; and
 - 2.2.4 not cause or allow Personal Data to be transferred outside the European Economic Area without the prior consent of NHS Supply Chain.
- 2.3 The Supplier and NHS Supply Chain shall ensure that Personal Data is safeguarded at all times in accordance with the Law, and this obligation will include (if transferred electronically) only transferring Personal Data (a) if essential, having regard to the purpose for which the transfer is conducted; and (b) that is encrypted in accordance with any international data encryption standards for healthcare, and as otherwise required by those standards applicable to NHS Supply Chain under any Law and Guidance (this includes, data transferred over wireless or wired networks, held on laptops, CDs, memory sticks and tapes).
- 2.4 For the avoidance of doubt, the Supplier must not process Sensitive Personal Data under or in connection with this Framework Agreement unless it has complied fully with:
 - 2.4.1 the requirements in Clauses 2.1 to 2.4 of this Schedule 3 (inclusive); and

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 36 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 2.4.2 the requirements of and its obligations under the Data Protection Act 1998 and The Data Protection (Processing of Sensitive Personal Data) Order 2000 (as amended from time to time) or any successor legislation.
- 2.5 Where any Personal Data is Processed by any Sub-contractor of the Supplier in connection with this Framework Agreement, the Supplier shall procure that such Sub-contractor shall comply with the relevant obligations set out in Clause 2 of this Schedule 3, as if such Sub-contractor were the Supplier.
- 2.6 The Supplier shall indemnify and keep NHS Supply Chain indemnified against, any loss, damages, costs, expenses (including without limitation legal costs and expenses), claims or proceedings whatsoever or howsoever arising from the Supplier's unlawful or unauthorised Processing, destruction and/or damage to Personal Data and/or Sensitive Personal Data in connection with this Framework Agreement.

3 Freedom of Information and Transparency

- 3.1 The Parties acknowledge the duties of Contracting Authorities under the FOIA, Codes of Practice and Environmental Regulations and shall give each other all reasonable assistance as appropriate or necessary to enable compliance with those duties.
- 3.2 The Supplier shall assist and cooperate with NHS Supply Chain to enable it to comply with its disclosure obligations under the FOIA, Codes of Practice and Environmental Regulations. The Supplier agrees:
 - 3.2.1 that this Framework Agreement and any recorded information held by the Supplier on NHS Supply Chain's behalf for the purposes of this Framework Agreement are subject to the obligations and commitments of NHS Supply Chain under the FOIA, Codes of Practice and Environmental Regulations;
 - 3.2.2 that the decision on whether any exemption to the general obligations of public access to information applies to any request for information received under the FOIA, Codes of Practice and Environmental Regulations is a decision solely for NHS Supply Chain;
 - 3.2.3 that where the Supplier receives a request for information under the FOIA, Codes of Practice and Environmental Regulations and the Supplier itself is subject to the FOIA, Codes of Practice and Environmental Regulations it will liaise with NHS Supply Chain as to the contents of any response before a response to a request is issued and will promptly (and in any event within two (2) Business Days) provide a copy of the request and any response to NHS Supply Chain;
 - 3.2.4 that where the Supplier receives a request for information under the FOIA, Codes of Practice and Environmental Regulations and the Supplier is not itself subject to the FOIA, Codes of Practice and Environmental Regulations, it will not respond to that request (unless directed to do so by NHS Supply Chain) and will promptly (and in any event within two (2) Business Days) transfer the request to NHS Supply Chain;
 - 3.2.5 that NHS Supply Chain, acting in accordance with the Codes of Practice issued and revised from time to time under both section 45 of FOIA, and

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7 August 2017 Page 37 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

regulation 16 of the Environmental Regulations, may disclose information concerning the Supplier and this Framework Agreement; and

- 3.2.6 to assist NHS Supply Chain in responding to a request for information, by processing information or environmental information (as the same are defined in FOIA and the Environmental Regulations) in accordance with a records management system that complies with all applicable records management recommendations and codes of conduct issued under section 46 of FOIA, and providing copies of all information requested by NHS Supply Chain within five (5) Business Days of that request and without charge.
- 3.3 The Parties acknowledge that, except for any information which is exempt from disclosure in accordance with the provisions of the FOIA, Codes of Practice and Environmental Regulations, the content of this Framework Agreement is not Confidential Information.
- 3.4 Notwithstanding any other term of this Framework Agreement, the Supplier consents to the publication of this Framework Agreement in its entirety (including variations), subject only to the redaction of information that is exempt from disclosure in accordance with the provisions of the FOIA, Codes of Practice and Environmental Regulations.
- 3.5 In preparing a copy of this Framework Agreement for publication under Clause 3.4 of this Schedule 3, NHS Supply Chain may consult with the Supplier to inform decision making regarding any redactions but the final decision in relation to the redaction of information will be at NHS Supply Chain's absolute discretion.
- 3.6 The Supplier shall assist and cooperate with NHS Supply Chain to enable NHS Supply Chain to publish this Framework Agreement.
- 3.7 Where any information is held by any Sub-contractor of the Supplier in connection with this Framework Agreement, the Supplier shall procure that such Sub-contractor shall comply with the relevant obligations set out in Clause 3 of this Schedule 3, as if such Sub-contractor were the Supplier.

4 Information Security

- 4.1 Without limitation to any other information governance requirements set out in this Schedule 3, the Supplier shall:
 - 4.1.1 notify NHS Supply Chain forthwith of any information security breaches or near misses (including without limitation any potential or actual breaches of confidentiality or actual information security breaches) in line with NHS Supply Chain's information governance Policies; and
 - 4.1.2 fully cooperate with any audits or investigations relating to information security and any privacy impact assessments undertaken by NHS Supply Chain and shall provide full information as may be reasonably requested by NHS Supply Chain in relation to such audits, investigations and assessments.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7 August 2017 Page 38 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

Schedule 4

Definitions and Interpretations

1 <u>Definitions</u>

1.1 In this Framework Agreement the following words shall have the following meanings unless the context requires otherwise, other than in relation to the Call-off Terms and Conditions for the Supply of Goods at Schedule 9 of this Framework Agreement. The definitions and Interpretations that apply to the Call-off Terms and Conditions for the Supply of Goods are as set out at Schedule 9 of this Framework Agreement.

"Action Plan"	shall have the meaning given to it in Clause 6 of Schedule 8;	
"Authority"	means the authority named on the Order Form;	
"Blue Diamond"	means a route of Supply whereby NHS Supply Chain (as the Authority) places an Order with the Supplier on behalf of an NHS Supply Chain customer, which is delivered by the Supplier to NHS Supply Chain for forward delivery onto the customer;	
"Business Continuity Event"	means any event or issue that could impact on the operations of the Supplier and its ability to fulfil its obligations under this Framework Agreement including an influenza pandemic and any Force Majeure Event;	
"Business Continuity Plan"	means the Supplier's business continuity plan which includes its plans for continuity of the supply of the Goods and Services during a Business Continuity Event;	
"Business Day"	means any day other than Saturday, Sunday, Christmas Day, Good Friday or a statutory bank holiday in England and Wales;	
"Call-off Terms and Conditions for the Supply of Goods"	means the call-off terms and conditions for Contracts as set out at Schedule 9 of this Framework Agreement forming part of the Contracts placed under this Framework Agreement;	
"Codes of Practice"	shall have the meaning given to the term in Clause 1.2 of Schedule 3;	
"Commencement Date"	16 April 2024;	
"Commercial means the document set out at Schedule 6; Schedule"		
"Confidential Information"	means information, data and material of any nature, which either Party may receive or obtain in connection with the conclusion and/or operation of the Framework Agreement including any procurement process which is:	
LEGAL TEMP 810-01 NHS	Supply Chain Framework Agreement for the Supply of Goods	
Revision: 7	August 2017 Page 39 of 90	

	 Personal Data or Sensitive Personal Data including without limitation which relates to any patient or other service user or his or her treatment or clinical or care history; 	
	 (b) designated as confidential by either party or that ought reasonably to be considered as confidential (however it is conveyed or on whatever media it is stored); and/or 	
	(c) Policies and such other documents which the Supplier may obtain or have access to through NHS Supply Chain's intranet;	
"Contract"	means any Contract entered into under this Framework Agreement with the Supplier by any Participating Authority as further defined in the Call-off Terms and Conditions for the Supply of Goods;	
"Contracting Authority"	means any contracting authority as defined in regulation 2 of the Public Contracts Regulations 2015 (as amended) (2015/102), other than NHS Supply Chain;	
"Contract Manager"	means for NHS Supply Chain and for the Supplier the individuals specified in the Key Provisions or such other person notified by a Party to the other Party from time to time in accordance with Clause 8.1 of Schedule 2;	
"Contract Price"	means the price exclusive of VAT that is payable to the Supplier by a Participating Authority under any Contract for the full and proper performance by the Supplier of its obligations under such Contracts (as calculated in accordance with the provisions of the Commercial Schedule) and as confirmed in the relevant Order Form relating to the particular Contract;	
"Data Protection Legislation"	means the Data Protection Act 1998 and any other Law relating to the protection of personal and sensitive personal data and the privacy of individuals, including where applicable guidance and codes of practice issued by the Information Commissioner;	
"Data Subject"	shall have the same meaning as set out in the Data Protection Act 1998;	
"Direct Route of Supply"	means a route of supply whereby the Authority (which is a Participating Authority who is not NHS Supply Chain) places an Order with the Supplier, which is delivered and invoiced directly to that Authority;	
"Dispute Resolution Procedure"	means the process for resolving disputes as set out in Clause 23 of Schedule 2;	
"DOTAS"	means the Disclosure of Tax Avoidance Schemes rules which require a promoter of tax schemes to tell HM Revenue & Customs of any specified notifiable arrangements or proposals	
LEGAL TEMP 810-01 NHS	S Supply Chain Framework Agreement for the Supply of Goods	
Revision: 7	August 2017 Page 40 of 90	
NHS Supply Ch	ain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)	

NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)

	and to provide prescribed information on those arrangements or
	proposals within set time limits as contained in Part 7 of the Finance Act 2004 and in secondary legislation made under vires contained in Part 7 of the Finance Act 2004 and as extended to National Insurance Contributions by the National Insurance Contributions (Application of Part 7 of the Finance Act 2004) Regulations 2012, SI 2012/1868 made under s.132A Social Security Administration Act 1992;
"Electronic Trading System(s)"	means such electronic data interchange system and/or world wide web application and/or other application with such message standards and protocols as NHS Supply Chain may specify from time to time;
"E-direct"	means Goods and Services ordered by NHS Supply Chain as the Authority on behalf of an NHS Supply Chain customer which are delivered directly to the customer and invoiced to NHS Supply Chain;
"Environment"	means the air, water, and land in or on which people, animals, and plants live;
"Environmental Regulations"	shall have the meaning given to the term in Clause 1.2 of Schedule 3;
"E-Procurement Guidance"	means the NHS E-Procurement Strategy available via:
	http://www.gov.uk/government/collections/nhs-procurement
	together with any further Guidance issued by the Department of Health in connection with it;
"Equality Legislation"	means any and all legislation, applicable guidance and statutory codes of practice relating to equality, diversity, non- discrimination and human rights as may be in force in England and Wales from time to time including, but not limited to, the Equality Act 2010, the Part-time Workers (Prevention of Less Favourable Treatment) Regulations 2000 and the Fixed-term Employees (Prevention of Less Favourable Treatment) Regulations 2002 (SI 2002/2034) and the Human Rights Act 1998;
"Ex Works"	means Goods and Services ordered from the Supplier based on the Contract Price, excluding delivery and other associated delivery costs, it being the responsibility of the Authority to arrange for collection of such Goods and Services from the Supplier;
"FOIA"	shall have the meaning given to the term in Clause 1.2 of Schedule 3;
"Force Majeure Event"	means any event beyond the reasonable control of the Party in question to include, without limitation:

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7 August 2017 Page 41 of 90				
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

	(a)	war including civil war (whether declared or undeclared), riot, civil commotion or armed conflict materially affecting either Party's ability to perform its obligations under this Framework Agreement;
	(b)	acts of terrorism;
	(c)	flood, storm or other natural disasters;
	(d)	fire;
	(e)	unavailability of public utilities and/or access to transport networks to the extent no diligent supplier could reasonably have planned for such unavailability as part of its business continuity planning;
	(f)	government requisition or impoundment to the extent such requisition or impoundment does not result from any failure by the Supplier to comply with any relevant regulations, laws or procedures (including such laws or regulations relating to the payment of any duties or taxes) and subject to the Supplier having used all reasonable legal means to resist such requisition or impoundment;
	(g)	compliance with any local law or governmental order, rule, regulation or direction that could not have been reasonably foreseen;
	(h)	industrial action which affects the ability of the Supplier to supply the Goods and/or Services, but which is not confined to the workforce of the Supplier or the workforce of any Sub-contractor of the Supplier; and
	(i)	a failure in the Supplier's and/or Authority's supply chain to the extent that such failure is due to any event suffered by a member of such supply chain, which would also qualify as a Force Majeure Event in accordance with this definition had it been suffered by one of the Parties;
"Framework Agreement"	means the form of framework agreement at the front of this document and all schedules attached to the form of framework agreement;	
"Fraud"	means any offence under any law in respect of fraud in relation to this Framework Agreement or defrauding or attempting to defraud or conspiring to defraud the government, parliament or any Contracting Authority;	
"General Anti-Abuse	means	
Rule"	(a)	the legislation in Part 5 of the Finance Act 2013; and

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7 August 2017 Page 42 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

	 (b) any future legislation introduced into parliament to counteract tax advantages arising from abusive arrangements to avoid national insurance contributions; 	
"Good Industry Practice"	means the exercise of that degree of skill, diligence, prudence, risk management, quality management and foresight which would reasonably and ordinarily be expected from a skilled and experienced supplier engaged in the manufacture and/or supply of goods and/or services similar to the Goods and/or Services under the same or similar circumstances as those applicable to this Framework Agreement, including in accordance with any codes of practice published by relevant trade associations;	
"Goods"	means all goods, materials or items that the Supplier is required to supply to Participating Authorities under Contracts placed under this Framework Agreement and/or made available for purchase under the Framework Agreement in accordance with Clause 22 of Schedule 2 and/or the Commercial Schedule, details of such Goods, materials or other items being set out in the Specification and Tender Response Document and any Order;	
"Group"	means in relation to a Party, that Party, any subsidiary or holding company from time to time of that Party, and any subsidiary from time to time of a holding company of that Party and holding company and subsidiary company shall have the meaning given in Section 1159 of the Companies Act 2006;	
"Guidance"	means any applicable guidance, direction or determination and any policies, advice or industry alerts which apply to the Goods and Services, to the extent that the same are published and publicly available or the existence or contents of them have been notified to the Supplier by NHS Supply Chain and/or have been published and/or notified to the Supplier by the Department of Health, Monitor, NHS England, the Medicines and Healthcare Products Regulatory Agency, the European Medicine Agency the European Commission, the Care Quality Commission and/or any other regulator or competent body;	
"Halifax Abuse Principle"	means the principle explained in the CJEU Case C-255/02 Halifax and others;	
"Intellectual Property Rights"	means all patents, copyright, design rights, registered designs, trademarks, know-how, database rights, confidential formulae and any other intellectual property rights and the rights to apply for patents and trademarks and registered designs;	
"Key Provisions"	means the key provisions set out in Schedule 1;	
Rey FIOVISIONS	"KPI" means the key performance indicators as set out in Schedu	
"KPI"	means the key performance indicators as set out in Schedule 8;	

Leave remained of this supply chain namework regreement of the supply of doods			
Revision: 7	August 2017	Page 43 of 90	
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

	 (a) any applicable statute or proclamation or any delegated or subordinate legislation or regulation as applicable in England and Wales; 	
	(b) any applicable European Union directive, regulation, decision or law;	
	 (c) any enforceable community right within the meaning of section 2(1) European Communities Act 1972; 	
	 (d) any applicable judgment of a relevant court of law which is a binding precedent in England and Wales; 	
	(e) requirements set by any regulatory body as applicable in England and Wales; and	
	(f) any applicable code of practice as applicable in England and Wales,	
	 (g) any relevant collective agreement and/or international law provisions (to include, without limitation, as referred to in (a) to (f) above). 	
"Management Fee"	has the meaning given under Clause 9.2 of Schedule 2;	
"Management Fee Report"	has the meaning given under Clause 9.3 of Schedule 2;	
"Monthly Service Level"	has the meaning given under Clause 3 of Schedule 8;	
"NHS"	means the National Health Service;	
"NHS Supply Chain's Obligations"	means NHS Supply Chain's further obligations, if any, referred to in the Specification and Tender Response Document;	
"Non-direct Route of Supply"	means all routes of supply through which NHS Supply Chain (as the Authority) places an Order with the Supplier for Goods and/or Services and the Supplier invoices NHS Supply Chain for the sum of the relevant Order, whether or not such Goods and/or Services are delivered to NHS Supply Chain or another authority and whether or not such Goods and/or Services are collected Ex Works. Non-direct routes of supply include E-Direct, Blue Diamond and Stock (and any other non-direct routes which NHS Supply Chain may notify to the Supplier from time to time);	
"Occasion of Tax Non- Compliance"	means:	
	 (a) any tax return of the Supplier submitted to a Relevant Tax Authority on or after 1 October 2012 is found on or after 1 April 2013 to be incorrect as a result of: 	

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7 August 2017 Page 44 of 90				
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

	i. a Relevant Tax Authority successfully challenging the Supplier under the General Anti-		
	Abuse Rule or the Halifax Abuse Principle or under any tax rules or legislation that have an effect equivalent or similar to the General Anti- Abuse Rule or the Halifax Abuse Principle;		
	 the failure of an avoidance scheme which the Supplier was involved in, and which was, or should have been, notified to a Relevant Tax Authority under the DOTAS or any equivalent or similar regime; and/or 		
	(b) any tax return of the Supplier submitted to a Relevant Tax Authority on or after 1 October 2012 gives rise, on or after 1 April 2013, to a criminal conviction in any jurisdiction for tax related offences which is not spent at the Effective Date or to a civil penalty for fraud or evasion;		
"Order Form"	means an order form on which Orders are to be placed, containing the details set out in Schedule 7;		
"Ordering Procedure"	means the procedure enabling Participating Authorities to call- off Goods and Services and enter into Contracts under this Framework Agreement, as set out in Schedule 7;		
"Orders"	means orders for Goods and Services placed under this Framework Agreement by Participating Authorities;		
"Organisation"	means the Supplier company;		
"Participating Authority"	means a Contracting Authority entitled to place Orders under this Framework Agreement including NHS Supply Chain and any other Contracting Authority as set out in the Key Provisions;		
"Party"	means NHS Supply Chain or the Supplier as appropriate and Parties means both NHS Supply Chain and the Supplier;		
"Personal Data"	means personal data as defined in the Data Protection Act 1998;		
"Policies"	means the policies, rules and procedures of NHS Supply Chain as notified to the Supplier from time to time;		
"Process" has the meaning given to it under the Data Protection Legisle and, for the purposes of this Framework Agreement, it include both manual and automatic processing. Processing Processed shall be construed accordingly;			
"Product Information" means information (including images) concerning the Grand Services as may be reasonably requested by NHS Su Chain and supplied by the Supplier to NHS Supply Cha			
LEGAL TEMP 810-01 NHS	Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7August 2017Page 45 of 90			
NHS Supply Chai	n Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

	accordance with Clause 21 of Schedule 2 for inclusion in NHS	
	Supply Chain's product catalogue from time to time;	
"Prohibited Acts"	has the meaning given under Clause 30.1.1 of Schedule 2;	
"Remedial Proposal"	has the meaning given under Clause 16.3 of Schedule 2;	
"Sensitive Personal Data"	means sensitive personal data as defined in the Data Protection Act 1998;	
"Services"	means any services which are ancillary to or associated with the Goods, which are purchased by Participating Authorities under Contracts placed under this Framework Agreement and/or made available for purchase under the Framework Agreement in accordance with Clause 22 of Schedule 2 and/or the Commercial Schedule, details of such Services being set out in the Specification and Tender Response Document and any Order;	
"Specification"	means the document set out in Schedule 5(a) as amended and/or updated in accordance with this Framework Agreement;	
"Staff"	means all persons employed or engaged by the Supplier to perform its obligations under this Framework Agreement including any Sub-contractors and person employed or engaged by such Sub-contractors;	
"Stock"	means Goods purchased by NHS Supply Chain (as an Authority) which are delivered and invoiced to NHS Supply Chain to be held as stock until such time as NHS Supply Chain customers place an order for such goods with NHS Supply Chain;	
"Sub-contract"	means a contract between two or more suppliers, at any stage of remoteness from the Supplier in a sub-contracting chain, made wholly or substantially for the purpose of performing (or contributing to the performance of the whole or any part of this Framework Agreement);	
"Sub-contractor"	means a party to a Sub-contract other than the Supplier;	
"Supplier"	means the supplier named on the form of Framework Agreement on the first page;	
"Tender Response Document"	means the document set out in Schedule 5(b) as accepted by NHS Supply Chain;	
"Term"	means the term as set out in the Key Provisions;	
"Third Party Body"	has the meaning given under Clause 8.5 of Schedule 2; and	
"VAT"	means value added tax chargeable under the Value Added Tax Act 1994 or any similar, replacement or extra tax.	

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 46 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 1.2 References to any statute or order shall include any statutory extension, modification or re-enactment, and any order, regulation, bye-law or other subordinate legislation.
- 1.3 References to any legal entity shall include any body that takes over responsibility for the functions of such entity.
- 1.4 References in this Framework Agreement to a "Schedule", "Appendix", "Paragraph" or to a "Clause" are to schedules, appendices, paragraphs and clauses of this Framework Agreement.
- 1.5 References in this Framework Agreement to a day or to the calculation of time frames are references to a calendar day unless expressly specified as a Business Day.
- 1.6 Unless set out in the Commercial Schedule as a chargeable item and subject to Clause 31.6 of Schedule 2, the Supplier shall bear the cost of complying with its obligations under this Framework Agreement.
- 1.7 The headings are for convenience only and shall not affect the interpretation of this Framework Agreement.
- 1.8 Words denoting the singular shall include the plural and vice versa.
- 1.9 Where a term of this Framework Agreement provides for a list of one or more items following the word "including" or "includes" then such list is not to be interpreted as an exhaustive list. Any such list shall not be treated as excluding any item that might have been included in such list having regard to the context of the contractual term in question. General words are not to be given a restrictive meaning where they are followed by examples intended to be included within the general words.
- 1.10 Where there is a conflict between the Supplier's responses to NHS Supply Chain's requirements set out in the Specification in the Tender Response Document and any other part of this Framework Agreement, such other part of this Framework Agreement shall prevail.
- 1.11 Where a document is required under this Framework Agreement, the Parties may agree in writing that this shall be in electronic format only.
- 1.12 Any guidance notes in grey text do not form part of this Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7 August 2017 Page 47 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

Schedule 5

5(a) Specification AIRWAY MANAGEMENT PRODUCTS AND ASSOCIATED EQUIPMENT

1. Introduction

- **1.1.** The Framework Agreement is for the supply of Airways Products and Associated Consumables and Accessories for the Airway Management of patients to ensure breathing and respiration.
- 1.2. The Framework Agreement is for the following Lots:

Lot Number	Lot Title
1	Laryngoscopes, Video Laryngoscopes, Tracheal Intubation Equipment,
	Single Use Upper Airway Devices, Single use Bronchoscopes and Single
	Use Flexible Intubating Endoscopes and Related Accessories
2	Endotracheal Tubes, Endobronchial Tubes and Blockers, Tracheostomy
	Tubes, Supraglottic Airways and Simple Airway Adjuncts
3	Breathing Systems Circuits and Accessories
4	Humidification and Filtration (HME's, HMEF's, Bacterial and Viral Filters)

- 1.3. Full technical specifications of the product lines awarded to the Framework Agreement (each a "Technical Specification" and together the "Technical Specifications") must be made available to NHS Supply Chain on request during the term of the Framework Agreement.
 - 1.3.1. Applicants must notify NHS Supply Chain immediately about any proposed changes to the Technical Specifications throughout the term of the Framework Agreement.
 - 1.3.2. If changes to the Technical Specification of any product line awarded to the Framework Agreement mean that the product line no longer meets the minimum requirements outlined in the Specification, NHS Supply Chain reserves the right to exclude that product line from the Framework Agreement.
 - 1.3.3. NHS Supply Chain reserves the right to request evidence of compliance with the Specification throughout the term of the Framework Agreement.
- 1.4. This Framework Agreement Specification makes reference to a number of standards and legislation. The list of standards and legislation is not intended to be exhaustive and any relevant standards and legislation which applies to the Framework Agreement (even if not stated) must be complied with by Applicants (together with those listed in this Framework Agreement Specification the "Standards and Legislation").
- **1.5.** Product lines must comply with the Standards and Legislation (as amended, extended or re-enacted from time to time).

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 48 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

1.6. Evidence of compliance to the Standards and Legislation must be provided by Applicants awarded to the Framework Agreement ("Suppliers") to NHS Supply Chain on request during the term of the Framework Agreement; in the event that sufficient evidence is not provided by Suppliers NHS Supply Chain reserves the right to suspend product lines until such evidence is provided by Suppliers.

2. Criteria applicable across all product lines

2.1. Standards and Legislation

	/here products are classed as Medical Devices as per the definition under edical Devices Regulation 2017/745 the following will apply:
AI	edical Devices Directive 93/42/EEC (as amended) Il products must have their CE or UKCA marking evident on the product and/or ackaging.
0	r
	edical Devices Regulation 2017/745 (as amended) Il products must have their CE marking evident on the product and/or packaging
	S EN ISO 20417:2021 (previously BS EN 1041:2008 +A1:2013.) edical devices. Information to be supplied by the manufacturer
Μ	S EN ISO 15223-1:2016 or BS EN ISO 15223-1:2021 edical devices. Symbols to be used with medical device labels, labelling and formation to be supplied.
Aı	S EN ISO 18190:2016 naesthetic and respiratory equipment. General requirements for airways and lated equipment.
w	here applicable products must conform to;
AI	nits of Measurement Directive 80/181/EEC Il products must have their CE marking clearly evident on the product and/or ackaging.
Aı	S ISO 18190:2016 naesthetic and respiratory equipment. General requirements for airways and elated equipment

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 49 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

BS EN ISO 18082:2014+A1:2017

Anaesthetic and respiratory equipment. Dimensions of non-interchangeable screw-threaded (NIST) low-pressure connectors for medical gases.

BS ISO 18562-1:2017

Biocompatibility evaluation of breathing gas pathways in healthcare applications. Evaluation and testing within a risk management process.

BS ISO 18562-2:2017

Biocompatibility evaluation of breathing gas pathways in healthcare applications. Tests for emissions of particulate matter.

BS ISO 18562-3:2017

Biocompatibility evaluation of breathing gas pathways in healthcare applications. Tests for emissions of volatile organic compounds (VOCs).

BS ISO 18562-4:2017

Biocompatibility evaluation of breathing gas pathways in healthcare applications. Tests for leachables in condensate.

BS ISO 18835:2015

Inhalational anaesthesia systems. Draw-over anaesthetic systems.

BS EN ISO 80369-1:2010

Small bore connectors for liquids and gasses in healthcare applications. General requirements.

BS EN ISO 80369-7:2017

Small-bore connectors for liquids and gases in healthcare applications. Connectors for intravascular or hypodermic applications.

BS EN ISO 5367:2014

Anaesthetic and respiratory equipment. Breathing sets and connectors.

BS EN ISO 5356-1:2015

Anaesthetic and respiratory equipment. Conical connectors. Cones and sockets

Where products are sterile, they must comply with either applicable standard below or equivalent international standard to designate device as sterile.

BS EN 556-1-2001

Sterilization of medical devices. Requirements for medical devices to be designated "STERILE". Requirements for terminally sterilized medical devices.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 50 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

BS EN556-2-2015

Sterilization of medical devices. Requirements for medical devices to be designated "STERILE". Requirements for aseptically processed medical devices. Where a product is sterilised an applicable validated sterilisation and routine control process must be applied; for example

BS EN ISO 14937:2009

Sterilization of health care Sterilization of health care products.

BS EN ISO 11135:2014+A1:2019

Sterilization of health-care products. Ethylene oxide

BS EN ISO 11137 series

Sterilization of health care products. Radiation

BS EN ISO 17665-1:2006

Sterilization of health care products. Moist heat. Requirements for the development, validation and routine control of a sterilization process for medical devices

- 2.2. On request applicants must provide NHS Supply Chain with Safety Data Sheets (SDS) for all products that fall under REACH (Registration, Evaluation, Authorisation and restriction of Chemicals) 2007 more specifically, an SDS must be provided if a substance or a mixture supplied is classified as hazardous under the CLP Regulation (EC) No 1272/2008.
- 2.3. If a product line contains phthalates this must be indicated on the packaging of that product line in accordance with Directive 2007/47/EC (amending Directives 90/385/EEC and 93/42/EEC).
- 2.4. If a product contains DEHP must be stated on the individual product packaging or IFU (instructions for use) and / or made available to NHS Supply Chain or end user on request.
- 2.5. Electrical product lines must comply with the requirements of the Directive on waste electrical and electronic equipment (WEEE Directive 2012/19/EU) and the Directive on the restriction of the use of certain hazardous substances in electrical and electronic equipment (RoHS 2 Directive 2011/65/EU).
- **2.6.** All product lines and packaging should be latex free where possible. If a product line or any packaging contains or does not contain latex this must be labelled on the product line or packaging (as applicable) to inform the user.
- **2.7.** During the term of the Framework Agreement Applicants must make NHS Supply Chain aware of any awarded product line that is classed by the MHRA as a Medicinal Product.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 51 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 2.8. All product line(s) must be supplied with a minimum 1 year's shelf life.
- **2.9.** All product lines must be delivered free of charge to a location as directed by either NHS Supply Chain or the customer.
- 2.10. IFUs must be written in English or pictograms and included on the individual product packaging and/or within the UOI and/or made available to NHS Supply Chain or end user on request.
- 2.11. Any cautions / warnings / contraindications to use must be provided in IFU.
- **2.12.** Must state details strictly necessary to identify the device for the user on the individual product packaging.
- 2.13. Lot number and expiry date must be stated on the individual product packaging.
- 2.14. For ordering purposes an identifier for example reference / manufacturing product code (MPC) must be stated on the individual product packaging and/or unit of issue packaging.
- 2.15. Product must be robust enough to resist breakage when used as directed by manufacturer.
- 2.16. Where applicable, products must be supplied sterile and individually wrapped.
- 2.17. Individual packaging must be of durable construction preventing product being pushed through, tampering and must not tear or rip apart during transportation and storage, to avoid damage to the product and / or breaches of product sterility and to reduce risk of plastic packaging being a foreign body when device being used.
- 2.18. Instructions for storage and disposal of device must be contained on the individual product packaging and/or IFU and / or made available to NHS Supply Chain or end user on request.
- **2.19.** Where applicable, the product packaging must include a non-adherent tab which allows product packaging to be opened at one end maintaining sterility.
- 2.20. Where applicable, the word single use and / or symbol must be depicted on the individual product packaging to inform the user of the products single use status in line with labelling (ISO 15223).
- **2.21.** Where applicable, products that are sterile, the transparent side of the individual packaging must allow visualisation of the contents.
- **2.22.** If MRI compatible it must be stated on the individual product packaging and / or unit of issue packaging.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	Revision: 7 August 2017 Page 52 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

- **2.23.** Where the product contains tubing, the packaging must be designed to minimise the risk of kinking of tubing while in storage.
- **2.24.** All breathing system components that have a requirement for a port with a cap: the cap must be tethered to the device.
- **2.25.** The product constituent's raw material/s must be made available to NHS Supply Chain or end user on request to support with customer recycling requirements.
- 2.26. Country of origin must be made available to NHS Supply Chain or end user on request.
- **2.27.** Weight of product must be made available to NHS Supply Chain or end user on request to provide trust with weight waste information.
- 2.28. External product packaging must be made of recyclable material where possible.
- 2.29. Where packaging can be recycled this must be displayed on the packaging.
- 2.30. Where a product is reusable, product must be able to be cleaned to prevent cross contamination and must have available cleaning / decontamination / re-sterilisation instructions. Instructions must be stated on the individual product packaging or IFU and/or made available to NHS Supply Chain or end user on request.
- 2.31. Training and implementation must be provided on request and in accordance with trust requirements through mutual agreement. This could be for example, providing ongoing support including pre-implementation education, supplier support and implementation guidance, bespoke clinical training, eLearning and post-implementation support.
- 3. Lot 1 Laryngoscopes, Video Laryngoscopes, Tracheal Intubation Equipment, Single Use Upper Airway Devices, Single use Bronchoscopes and Single Use Flexible Intubating Endoscopes and Related Accessories
- **3.1.** This Lot is for the supply of airway equipment to aid tracheal intubation. These can be supplied sterile or non-sterile. Products within this Lot include:

3.2. Laryngoscopes and Associated Accessories

- 3.2.1. Complete Laryngoscopes Systems.
- 3.2.2. Laryngoscope Handles.
- 3.2.3. Laryngoscope Blades.
- 3.2.4. Laryngoscope Accessories.

3.3. Video Laryngoscopes and Related Accessories

- 3.3.1. Integrated (portable) Single Use Video Laryngoscope Systems.
- 3.3.2. Integrated (portable) Reusable Video Laryngoscope Monitor and Camera.
- 3.3.3. Non-Integrated Video Laryngoscope.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7 August 2017 Page 53 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			

- 3.3.4. Standalone (cart based) Reusable Monitor.
- 3.3.5. Interchangeable Single Use Blades or Sheaths with or without a channel.
- 3.3.6. Interchangeable Reusable Blades or Sheaths with or without a channel.
- 3.3.7. Accessories which include replacement parts for the Video Laryngoscopes and items of equipment that are required in order for the Video Laryngoscope to be used e.g., Monitors, Modules, Battery Packs, Carry Cases, Charging Units, Connection Cables, Imagers, Batons, Wands, Blades, Stats, Stylets, Pole Mounts, Trolleys, Protection Caps, Stands, User Manuals.
- 3.3.8. Bougies.
- 3.3.9. Stylets.
- 3.3.10. Airway Exchange Catheters.
- 3.3.11. Single Use Flexible Intubating Endoscope and Related Accessories.
- 3.3.12. Single Use Upper Airway Devices and Single Use Bronchoscopes and Related Accessories.
- 3.3.13. Equipment for Difficult or Emergency Airway Management.

3.4. Standards and Legislation

STANDARD AND LEGISLATION

BS EN ISO 7376:2020

Anaesthetic and respiratory equipment. Laryngoscopes for tracheal intubation

3.5. Laryngoscopes and Associated Accessories - Laryngoscopes are devices used to perform direct laryngoscopy and to aid in tracheal intubation. They are used to visualise the larynx or pharynx for suctioning, removal of foreign body, placing of nasogastric tubes and throat packs. They can be single use, reusable, or reusable with single use blades. Laryngoscopes are handled medical instruments with interchangeable blades of different sizes used within the throat to enable access to the vocal cords and glottis and to aid intubation.

3.6. All Laryngoscopes must meet the following:

- 3.6.1. Product must be robust enough to resist breakage when assembling/dismantling as directed by manufacturer.
- 3.6.2. Ergonomic grip design to allow end users to have a firm grip to prevent patient harm- i.e., doesn't allow hand to slip.
- 3.6.3. Must provide a light source to allow a direct laryngoscopy to aid tracheal intubation alongside other desired effects such as visualisation of the larynx, pharynx for suctioning and removal of foreign bodies amongst other requirements.
- 3.6.4. Material type must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs and / or data sheets.
- 3.6.5. Compatibility colour coding must comply with BS EN ISO 7376:2020
- 3.6.6. Where products require to be assembled, the packaging must state type of light source to ensure avoidance of mismatch for example, fibreoptic.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 54 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 3.6.7. The device must have the ability to accept range of blades in accordance with the BS EN ISO 7376:2020.
- **3.7. Complete Laryngoscope System** consists of handle and integral blade. They can be various sizes and types of blades.
- **3.8. Laryngoscope Handles** A clean, non-sterile, hand-held segment of a laryngoscope (i.e., non-endoscopic rigid type) designed to manipulate the laryngoscope blade during airway access and intubation. It is a cylindrical device that contains batteries / light-emitting cells to provide the energy / light for the airway-illuminating feature of the laryngoscope (e.g., small light bulb, fibreoptics), and can accommodate various blades. It is available in various sizes.. They can be single use or reusable.
- 3.9. Laryngoscope Blades The segment of a laryngoscope (i.e., rigid intubation type) intended to be inserted into the oral cavity to manipulate the tongue, preventing it from obstructing the oropharynx and enabling a clear view of the trachea for the insertion of an endotracheal (ET) tube prior to the delivery of inhalation anaesthesia and / or ventilation. It is connected to the laryngoscope handle and illumination of the airway is provided by a small built-in light bulb or a fibreoptic light. This device may be curved or straight and of various designs and lengths; it may be hinged / interchanged with the handle.
- **3.10.** Accessories which includes equipment that is required to operate the laryngoscopes such as batteries, recharging units, battery/light containers, or bulbs.
- 3.11. Video Laryngoscopes, Single Use Flexible Intubation Endoscopes and Related Accessories - A collection of video laryngoscopic devices intended to be used for various procedures on the airway and respiratory tract including intubation, lung dual-chamber positioning, suction, biopsy, drug administration and oxygen giving. It typically consists of intubation laryngoscope blades, а set of video а video intubation laryngoscope handle/monitor, video а rigid non-bladed intubation laryngoscope and a flexible video intubation laryngoscope. They can have the ability to record to support teaching and clinical audit.
 - 3.11.1. Integrated (portable) Single Use Video Laryngoscope Systems An assembly of devices enabling video imaging of the trachea during insertion of an endotracheal (ET) tube. This is a single use device.
 - 3.11.2. Integrated (portable) Reusable Video Laryngoscope Monitor and Camera An assembly of devices enabling video imaging of the trachea during insertion of an endotracheal (ET) tube. This is a reusable device.
 - 3.11.3. **Non-Integrated Video Laryngoscope** An assembly of devices enabling video imaging of the trachea during insertion of an endotracheal (ET) tube. This can be a single use or reusable device.
 - 3.11.4. Standalone (cart based) Reusable Monitor A monitor for the user to view anatomical images of the larynx / trachea transmitted by a video

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 55 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

system from an electronic camera at the distal end of the blade. This is a reusable device.

- 3.11.5. Interchangeable Single Use Blades or Sheaths with or without a channel A sterile, rigid blade-like component of a bladed video intubation laryngoscope intended to be inserted into the oral cavity to manipulate the tongue, preventing it from obstructing the oropharynx and enabling a clear view of the trachea for the insertion of an endotracheal (ET) tube prior to the delivery of inhalation anaesthesia and/or ventilation. This is a single use device.
- 3.11.6. Interchangeable Reusable Blades or Sheaths with or without a channel A sterile, rigid blade-like component of a bladed video intubation laryngoscope intended to be inserted into the oral cavity to manipulate the tongue, preventing it from obstructing the oropharynx and enabling a clear view of the trachea for the insertion of an endotracheal (ET) tube prior to the delivery of inhalation anaesthesia and/or ventilation. This is a reusable device.
- 3.11.7. Accessories which include replacement parts for the video laryngoscopes and items of equipment that are required in order for the video laryngoscope to be used, for example, modules, battery packs, carry cases, charging units, connection cables, docking stations, bulbs, imagers, batons, wands, blades, stats, stylets, pole mounts, trolleys, protection caps, stands, user manuals.
- **3.12. Video Laryngoscopes** can vary dependent upon their intended use and some designs are unique, as a minimum they must include the following features:
 - 3.12.1. Ergonomic grip design to allow end users to have a firm grip to prevent patient harm- i.e. doesn't allow hand to slip.
 - 3.12.2. Provide a stable image to allow visualisation of the required anatomy.
 - 3.12.3. The image must be able to be viewed under a variety of clinical settings.
 - 3.12.4. Battery powered versions must run for a minimum of 2 hours on a set of fully charged batteries.
 - 3.12.5. Must have the ability to image capture, video capture and export data as required.
- **3.13. Bougies (Endotracheal Tube Introducers)** are designed as an intubation tool; The tracheal tube introducer, known as the bougie, is typically used to aid tracheal intubation in poor laryngoscopic views or after intubation attempts fail. They can be single use or reusable. They must have the following features:
 - 3.13.1. Material type must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs and / or data sheets.
 - 3.13.2. Size and length must be stated on the individual product packaging and / or unit of issue packaging and/or IFUs.
 - 3.13.3. Length graduation marks must be present on the device.
 - 3.13.4. Must have an atraumatic tip to minimise risk of patient harm.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Good					
Revision: 7	August 2017	Page 56 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)					

- 3.13.5. Must have a smooth exterior to minimise risk of patient harm and facilitate insertion.
- **3.14. Stylets** are a malleable metal wire designed for pre-forming the ET tube to a desired shape to facilitate intubation. The product line must have the following features:
 - 3.14.1. Must be designed to prevent the distal stylet tip from extending beyond the tip of the ET tube.
 - 3.14.2. Material type must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs and / or data sheets.
 - 3.14.3. Material must allow the ET tube shape to be altered to suit the individual patient.
 - 3.14.4. Material must allow the ET tube to be stiffened to aid passage into the trachea.
 - 3.14.5. Size and length must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs.
 - 3.14.6. Compatibility with airway devices must be stated on the individual product packaging and /or unit of issue packaging and / or IFUs.

3.15. Airway Exchange Catheters - are specifically used for ET tube exchange and have oxygenating adaptors. They must have the following features:

- 3.15.1. Must have visible depth markers.
- 3.15.2. Adaptors must be supplied with device to provide oxygen.
- 3.15.3. Must have an atraumatic tip to minimise risk of patient harm.
- 3.15.4. Must have a smooth exterior to minimise risk of patient harm and facilitate insertion.
- 3.15.5. Material type must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs and / or data sheets.
- 3.15.6. Size and length must be stated on the individual product packaging and/ or unit of issue packaging and / or IFUs.
- 3.15.7. Compatibility with airway devices must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs.
- 3.15.8. These are single use.
- **3.16. Intubation Catheter** designed for use with a fibreoptic bronchoscope to facilitate endotracheal tube exchange.
 - 3.16.1. Must have visible depth markers.
 - 3.16.2. Adaptors must be supplied with device to provide oxygen.
 - 3.16.3. Must have an atraumatic tip to minimise risk of patient harm.
 - 3.16.4. Must have a smooth exterior to minimise risk of patient harm and facilitate insertion.
 - 3.16.5. Material type must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs and / or data sheets.
 - 3.16.6. Size and length must be stated on the individual product packaging and / or unit of issue packaging and/or IFUs.
 - 3.16.7. Compatibility with airway devices must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs.

LEGAL TEMP 810	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	Page 57 of 90				
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)					

- **3.17. Single Use Flexible Intubation Video Endoscopes** designed for difficult airway procedures to visualise the airway. They connect to a live video monitor to give a clear view of the airway. As a minimum they must have the following features:
 - 3.17.1. Ergonomic grip design to allow end users to have a firm grip to prevent patient harm i.e., doesn't allow hand to slip.
 - 3.17.2. Must provide suction capability.
 - 3.17.3. Must be provided sterile.
 - 3.17.4. Must connect to a standalone monitor.
 - 3.17.5. The compatible monitor must have features to enable adjustment to suit different clinical settings.
 - 3.17.6. Must have the ability to image capture, video capture and export data as required.
- 3.18. Single Use Upper Airway Devices and Single Use Bronchoscopes and Related Accessories Single use upper airway devices and single use bronchoscopes are used in diagnostic and therapeutic interventions for a variety of upper airways and pulmonary disorders, they can also be used to visualise the anatomy in unexpected difficult airways. Accessories include associated equipment for the device such as monitors, cables and compatible devices. They must meet the following:
 - 3.18.1. Must have a working channel, if size appropriate, to be used with endoscopy instruments used in diagnostic or therapeutic procedures.
 - 3.18.2. Must have the ability to provide lavage and suction to support therapeutic interventions where applicable.
 - 3.18.3. Must connect to a monitor to visualise the anatomy.
 - 3.18.4. Product must achieve a neutral position to allow the clinician to accurately identify airway anatomy. The tip must be able to provide a full range of movement to visualise the anatomy through flex and retroflex movements. Product must maintain position until the clinician dictates required movement.
 - 3.18.5. Ergonomic grip design to allow end users to have an optimal grip to prevent patient and user harm- i.e. doesn't allow hand to slip, allows effective manipulation.
 - 3.18.6. Device must be lightweight in order to minimise risk of endoscopy-related injury.
 - 3.18.7. Provide a stable image to allow visualisation of the required anatomy
 - 3.18.8. The image must be able to be viewed under a variety of clinical settings.
 - 3.18.9. The device and/or associated accessories must have the ability to record, store, export and capture images to assist healthcare providers with auditing and training purposes.
 - 3.18.10. Compatibility with airway devices, suctioning devices, endoscopic accessories and specimen collection devices must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs.
 - 3.18.11. Accessories for single use Bronchoscopes.

LEGAL TEMP 810-01	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 58 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)					

3.19. Equipment for Difficult or Emergency Airway Management – Airway equipment that is required in an emergency situation as for "Cant Intubate, Cant Oxygenate" (CICO) situations and difficult airway management situations. This can include sets which include more than one of the product lines contained within this Framework in convenient to use packs to support management of a CICO or difficult airway management.

4. Lot 2 – Endotracheal Tubes, Endobronchial Tubes and Blockers, Tracheostomy Tubes, Supraglottic Airways and Simple Airway Adjuncts.

- **4.1.** This Lot is Airway Devices that provide a means of access and securement of an airway. This Lot includes:
 - 4.1.1. Standard Endotracheal Tubes; Cuffed and Non-Cuffed variants.
 - 4.1.2. Subglottic Endotracheal Tube; Cuffed and Non-Cuffed variants.
 - 4.1.3. Reinforced Endotracheal Tubes; Cuffed and Non-Cuffed variants.
 - 4.1.4. Preformed Endotracheal Tubes; Cuffed and Non-Cuffed variants.
 - 4.1.5. Cole Tracheal Tube.
 - 4.1.6. Speciality Endotracheal Tube; Cuffed and Non-Cuffed variants.
 - 4.1.7. Endotracheal Tube Holders.
 - 4.1.8. Endobronchial Tubes:
 - 4.1.8.1. Single Lumen Endobronchial Tubes (SLT).
 - 4.1.8.2. Double Lumen Endobronchial Tubes, (DLT).
 - 4.1.8.3. Video Double Lumen Endobronchial Tubes (VDLT).
 - 4.1.8.4. Video Single Lumen Endobronchial Tubes.
 - Endobronchial Blockers (BB).
 - 4.1.10. Endotracheal and Endobronchial Accessories.
 - 4.1.11. Tracheostomy Tubes:

4.1.9.

- 4.1.11.1. Cuffed Tracheostomy Tubes.
- 4.1.11.2. Cuffed Subglottic Tracheostomy Tubes.
- 4.1.11.3. Fenestrated Tracheostomy Tubes (Cuffed and Uncuffed).
- 4.1.11.4. Single Lumen Tracheostomy Tubes.
- 4.1.11.5. Double Lumen Tube (Tube and Inner Cannula/Coaxial).
- 4.1.11.6. Adjustable Flange Tracheostomy Tubes.
- 4.1.11.7. Tracheostomy Inner Cannula.
- 4.1.11.8. Speciality Tubes.
- 4.1.12. Laryngectomy Tubes.
- 4.1.13. Laryngectomy Inner Cannula.
- 4.1.14. Tracheostomy Tube Holders.
- 4.1.15. Tracheostomy Fasteners/Ties.
- 4.1.16. Percutaneous Single Stage Dilation Sets.
- 4.1.17. Tracheostomy Tube Accessories.
- 4.1.18. Standard Laryngeal Masks.
- 4.1.19. Reinforced Laryngeal Masks.
- 4.1.20. Intubating Laryngeal Masks.
- 4.1.21. Preformed Laryngeal Masks.
- 4.1.22. Speciality Laryngeal Masks.

LEGAL TEMP 810-0	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	August 2017	Page 59 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 4.1.23. Additional Supraglottic Devices Additional supraglottic devices can incorporate double cuffed tubes for example.
- 4.1.24. Non inflatable Cuffed Supraglottic Device A device where the cuff is not required to be inflated and can be inserted into the airway where it forms to the patient's anatomy.
- 4.1.25. Simple Airway Adjuncts.
- 4.1.26. Oropharyngeal Airways.
- 4.1.27. Nasopharyngeal Airways (NPA).
- 4.1.28. Supraglottic and Airway Adjuncts Accessories.
- 4.2. Endotracheal Tubes are a hollow cylinder inserted orally or nasally to provide a means of securing the patient's airway, allowing spontaneous and controlled ventilation. It is available in various diameters and lengths for adult and paediatric patients. These can be single use or reusable. They can be cuffed or uncuffed. They can be straight or curved.

4.3. Standards and Legislation – All Tracheal and Endotracheal Tubes and Connectors must conform to the below standards and legislation.

STANDARD AND LEGISLATION

BS EN ISO 5361:2016

Anaesthetic and respiratory equipment. Tracheal tubes and connectors

BS EN ISO 80369-1:2018

Small-bore connectors for liquids and gases in healthcare applications

4.4. Standard Endotracheal Tubes (ET) - must meet the following:

- 4.4.1. Must include a radio opaque line if not reinforced.
- 4.4.2. Must have indicated length of use on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
- 4.4.3. Must maintain its intended shape when removed from the manufacturers packaging.
- 4.4.4. Tracheal tube material must have sufficient rigidity with the thinnest possible wall whilst providing flexibility and resistance to collapse and kinking. NHS SC reserve the right to request the outcome of the test methods used in accordance with BS EN ISO 5361:2016.
- 4.4.5. Have its internal diameter marked on the outside of the tube in millimetres.
- 4.4.6. Have its outer diameter marked on the tube and packaging.
- 4.4.7. The tube must be able to be cut to size to suit the individual patient with the exception of any speciality tube not designed to be cut.
- 4.4.8. The length, in centimetres from the tip of the tube must be marked on the outside of the tube.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 6o of 9o		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 4.4.9. Must have black intubation depth marker(s) to assist in the placement of the tracheal tube within the trachea (vocal cord marker).
- 4.4.10. The bevel must be atraumatic to help prevent patient harm.
- 4.4.11. The bevel must be left facing unless other configurations are required as per clause 5.4.2 in BS EN ISO 5361.
- 4.4.12. Material type must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
- 4.4.13. Must be provided Sterile.
- 4.4.14. Can be provided with or without a side hole(s) also known as Murphy Eye(s).
- 4.4.15. If Murphy Eye(s) present:
 - 4.4.15..1. The Murphy Eye(s) must have a smooth surface to minimise trauma to the patient.
 - 4.4..15..2. The Murphy Eye must be on the opposite side of the bevel. The Murphy Eye must be designed in such a manner to ensure that the patient end of the tube is not unduly prone to kinking/collapse.

4.5. For Cuffed variants - the cuff must:

- 4.5.1. Provide an airtight seal between the tube and the tracheal wall.
- 4.5.2. Cuffs can be high volume / low pressure or low volume / high pressure.
- 4.5.3. The cuff resting diameter must be indicated on the tube and / or packaging and / or IFUs.
- 4.5.4. The IFUs must contain the recommended maximum cuff inflation pressure.
- 4.5.5. For cuffed variants there it must have a pilot balloon and / or other device such as a pressure indicator to indicate inflation / deflation of the cuff.
- 4.5.6. The pilot balloon / device must indicate whether the cuff is inflated or not.
- 4.5.7. The pilot balloon / device must be capable of being inflated and deflated multiple times.
- 4.5.8. Must have a self-sealing valve for injecting air.
- 4.5.9. The insertion point of the pilot balloon / device inflation lumen (tubing) must not affect the ability to cut the tube to a desired length as required.
- **4.6. Subglottic Endotracheal Tube (ET)** A tracheal tube which has a subglottic suction; is a small hole in the shaft of the tube, just above the cuff. A channel runs up inside the endotracheal tube and is connected to a suction port to remove secretions that pool above the cuff.
 - 4.6.1. Must meet all the requirements of paragraph 4.4 and 4.5 above.
 - 4.6.2. Must have the ability to provide subglottic aspiration to pooled secretions from above the cuff of a tracheal tube to help prevent ventilator-associated pneumonia (VAP).
 - 4.6.3. The lumen dedicated for suctioning must be designed in such a manner that supports prevention of occlusion.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 61 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 4.6.4. The subglottic suction port line must be marked as different to the endotracheal pilot cuff line.
- 4.6.5. Must have the ability to flush the aspiration port and associated line.
- 4.7. Reinforced Endotracheal Tubes (ET) are designed to resist kinking or compression. These tubes are frequently used during facial surgeries and neurosurgeries, and in patients in a non-supine position during surgery. Reinforced tracheal tubes contain a metal wire coil embedded in the wall of the tube shaft which keeps the lumen of the tube open when it is bent.
 - 4.7.1. Must meet all the requirements of paragraph 4.4 above and if cuffed meet 4.5 also.
 - 4.7.2. The metal wire must be embedded in the wall of the tube shaft.
 - 4.7.3. The wire reinforcement must be continuous along the length of the tube to minimise the risk of kinking.
- 4.8. Preformed Endotracheal (ET) Tube have a manufactured pre-formed shape to reduce the risk of kinking and obstruction and facilitate surgical access for intra oral and facial surgery. They can be "North Facing" or "South Facing". They can be reinforced or nonreinforced.
 - 4.8.1. Must meet all the requirements of paragraph 4.4 above and if cuffed meet 4.5 also.
 - 4.8.2. They must have a black indication mark at the point of the maximum angle of the bend.
- **4.9. Cole Endotracheal (ET) Tube** A Cole ET tube is a tube designed for neonates and paediatric patients due to its narrowing of the tube at the point where is passes through the larynx.
- 4.10. Speciality Endotracheal (ET) Tube Speciality ET tubes are designed to maintain an airway for a variety of procedures. They include ET Tubes such as laser resistant, integrated throat / pharyngeal pack, Microlaryngoscopy Tube, Compatible ET for supraglottic devices and Evoked Potentials tracheal tubes.
- 4.11. Endotracheal Tube Holder is a device designed to hold and stabilize an endotracheal (ET) tube when it has been intubated into a patient; it may also include a teeth protector to prevent damage to the teeth or tube during intubation. This is a single-use device. Available in adult, paediatric and neonatal sizes.
 - 4.11.1. The holder must secure the airway device to minimise the risk of accidental extubation or displacement.
 - 4.11.2. The device must have features or methods to rapidly disconnect ET tube in the event of an emergency.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods				
Revision: 7	August 2017	Page 62 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 4.11.3. The device must be designed that will support securement under expected conditions such as repositioning of the holder, patient secretions, patient transport.
- 4.11.4. The holder must enable adjustment to accommodate individual patient sizes.
- 4.11.5. The holder must enable securement of the ET tube and the relevant size(s) this must be stated on the individual product packaging.
- 4.11.6. The holder can incorporate a bite block.
- 4.11.7. Material type must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs and / or data sheets.
- 4.11.8. Must be designed to minimise discomfort and potential harm to the patient particularly around known pressure areas.
- 4.11.9. The ET tube holder must work in conjunction with the ET tube to enable continuous securement of a definitive airway.
- **4.12. Endobronchial Tubes** A hollow cylinder that is designed for oral insertion via the trachea and into a lung to maintain airway patency and/or to deliver anaesthetic inhalation agents or other medical gases, and secure ventilation. It is typically made of plastic, rubber, or silicone and may be packaged with a connector that attaches to a breathing circuit and is available in two basic configurations: 1) a double lumen tube with independent cuffs at the distal end for ventilation of the left or right bronchi; or 2) a twin lumen tube with independent cuffs for selective lung ventilation during thoracic surgery or postoperative care when independent lung ventilation is indicated. This is a single-use device. All Endobronchial tubes must meet the following:
 - 4.12.1. The bronchial cuff and its pilot balloon inflation must be indicated for visual identification.
 - 4.12.2. Details for the size and use of the devices must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs, e.g., side specific use (side specific use relates to which side of the lung the product must be used on i.e. (left or right).
 - 4.12.3. Products within this Lot must be supplied sterile.
- **4.13. Single Lumen Endobronchial Tubes (SLT)** are advanced into the main bronchus of the non-operative lung to selectively ventilate it, while permitting slow collapse of the contralateral lung.
- **4.14. Double Lumen Endobronchial Tubes (DLT)** are bifurcated tube with separate tracheal and endobronchial lumens, that can be used to ventilate either lung independently. The tracheal lumen is designed to terminate above the carina, while the bronchial lumen is angled to fit into the appropriate main-stem bronchus.
- **4.15. Video Double Lumen Endobronchial Tubes (VDLT)** have an integrated camera and light source between the tracheal and bronchial cuffs, thus enabling continuous airway visualization during positioning of the tube.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods					
Revision: 7	August 2017	Page 63 of 90			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)					

- **4.16. Video Single Lumen Endobronchial Tubes** have an embedded camera and light source between the tracheal and bronchial cuffs, thus enabling continuous airway visualization during positioning of the tube.
- **4.17. Endobronchial Blockers (BB)** occlude the main-stem bronchus, thereby preventing ventilation distal to the occlusion. They can also be used to provide selective lobar collapse as well. They must meet the following:
 - 4.17.1. Details for the size and use of the devices must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs, e.g., side specific use (side specific use relates to which side of the lung the product must be used on i.e. (left or right).
- **4.18** Endotracheal and Endobronchial Accessories which include any product lines that would be used to connect a tube to another piece of equipment, bite guards for protection during product use, products for measuring and inflating cuffs; and Sets which include more than one of the relevant product lines contained within this Lot compiled in convenient to use.
- **4.18. Tracheostomy Tubes and Related Accessories** A tracheostomy tube is a sterile hollow arc shaped cylinder designed to be inserted directly into the trachea via a hole made into the neck allowing a patient to breathe without using their nose or mouth. It is available in various diameters and lengths for adults and paediatric patients. They can be cuffed or uncuffed and can be fenestrated or unfenestrated. They must be single patient use.
- 4.19. Standards and Legislation All Tracheostomy Tubes and Connectors must conform to the below standards and legislation

STAN					SLATI	ON			
BS EN	IIS	SO 5	366:2	2016					

Anaesthetic and respiratory equipment. Tracheostomy tubes and connectors

BS EN 1282-2:2005+A1:2009

Tracheostomy tubes. Paediatric tubes

- **4.20. Tracheostomy Tubes** All Tracheostomy tubes must meet the following general requirements:
 - 4.20.1. Must be single patient use.
 - 4.20.2. Size and length must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs.
 - 4.20.3. Material type of product components must be stated on the individual product packaging and/or unit of issue packaging and /or IFUs and /or data sheets.
 - 4.20.4. Must be stated if supplied with an obturator to aid insertion of device.

LEGAL TEMP 810-01	LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7	August 2017 Page 64 of			
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)				

- 4.20.5. Device must have means of securing tracheostomy to the neck for stabilisation for example Neck flange / base plate.
- 4.20.6. Securement device must have tracheostomy tube size, brand, cuff type, outer diameter in mm and the inner diameter in mm on the product.
- 4.20.7. Include a standard 15mm male connector at end to enable connection to ventilator circuit, resuscitation bags, speaking valves and caps.
- 4.20.8. The size of the tracheostomy tubes without inner cannulas must be stated according to the functional internal diameter (ID) at the narrowest point. This would relate to the part of the tube which has the 15mm connector fitted to it. This information will be present on the device, the individual product packaging and / or unit of issue packaging and / or IFUs.
- 4.20.9. For double cannula tracheostomy tubes, the inner diameter of the tracheostomy tube with the inner cannula fitted must be stated. This information will be present on the device, the individual product packaging and / or unit of issue packaging and /or IFUs.
- 4.20.10. Information regard to sizing, features and benefits of the devices and associated products must be made available to NHS Supply Chain on request.
- 4.20.11. Must be supplied sterile.

4.21. Cuffed Tracheostomy Tubes - must meet all previous points in 4.20 plus the following:

- 4.21.1. The tracheostomy cuff must be connected to the pilot balloon.
- 4.21.2. The inflated cuff on a cuffed tracheostomy tube must provide an airtight seal when used in accordance with manufacturer's instructions.
- 4.21.3. Cuffed tracheostomy tubes must have a pilot balloon and / or other device such as a pressure indicator to indicate inflation / deflation of the cuff.
- 4.21.4. Device must have ability to connect to a cuff pressure monitoring / recording device to enable measurement of the pressure within the inflated cuff to prevent pressure on the tracheal mucosa which can lead to pressure necrosis.
- 4.21.5. The pilot balloon must indicate whether the cuff is inflated or deflated.
- 4.21.6. The pilot balloon must be capable of being inflated and deflated numerous times.
- 4.21.7. The pilot balloon must have a self-sealing valve for injecting air into the cuff.
- 4.21.8. The IFUs must contain the recommended maximum cuff inflation pressure.
- 4.21.9. The cuff resting diameter must be indicated on the tracheostomy tube and/or packaging and/or IFUs.

4.22. Cuffed Subglottic Tracheostomy Tubes

4.22.1. Must meet all of 4.20 and 4.21.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 65 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 4.22.2. Must have the ability to provide subglottic aspiration to pooled secretions from above the cuff of a tracheal tube to help prevent ventilator-associated pneumonia (VAP).
- 4.22.3. Must have the ability to flush the aspiration port and associated line.
- 4.22.4. The lumen dedicated for suctioning must be designed in such a manner that supports prevention of occlusion.
- 4.22.5. The subglottic suction port line must be marked as different to the tracheostomy pilot cuff line.
- **4.23. Fenestrated Tracheostomy Tube** has a hole in the curvature in the posterior wall of the tracheostomy tube. Designed to allow airflow through the fenestration for better voicing, can be cuffed or uncuffed. Must meet all 4.20 plus the following:
 - 4.23.1. The fenestration must have a surface designed that minimises trauma to the patient.
 - 4.23.2. The fenestration must be designed in such a manner to ensure that the patient end of the tube is not unduly prone to kinking/collapse.
 - **4.24. Single Lumen Tracheostomy Tubes** consist of only an outer tube and have no inner cannula. Must meet all 4.20 plus the following:
 - 4.24.1. Material type must not detrimentally effect removal of secretions/prevent blocking of tube
- **4.25. Double Lumen Tracheostomy Tube** consists of an outer tracheostomy tube and an inner cannula, also known as coaxial tubes. These are used for one lung ventilation. Must meet all 4.20 plus the following:
 - 4.25.1. Inner cannula must connect securely to the tracheostomy tube to prevent accidental dislodgement.
 - 4.25.2. Single patient use inner cannulas must be able to be inserted and reinserted multiple times and remain secure once reinserted.
 - 4.25.3. Information regarding the number of times an inner cannula can be removed, inserted and cleaned must be stated on the UOI / Individual packaging and /or IFUs.
 - 4.25.4. Compatibility with correct tracheostomy for example, sizing, brand must be stated on the individual product packaging.
- **4.26. Adjustable Flange Tracheostomy tubes** are designed for patients with a deep-set trachea. The tracheostomy tube can be adjusted to the required length, when in-situ the flange can be moved to assist with cleaning. Must meet all 4.20 plus the following:
 - 4.26.1. Once adjusted the flange must be secure and minimise risk to patient harm.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 66 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- **4.27. Tracheostomy Inner Cannulas (Tubes)** A sterile, hollow, cylindrical, curved device designed as an internal, independent component of a tracheostomy tube. It has precise dimensions allowing it to fit exactly into the outer tube and function as the inner cannula for tracheostomy tubes that have this kind of function. It is removed once or more times daily to facilitate the removal of phlegm, slime, and contamination build-up, thereby preventing blockage and infection of the stoma and eliminates the traumatic removal of the tracheostomy tube. There are a variety which can be single patient use or single use.
 - 4.27.1. Compatibility with correct tracheostomy for example, sizing, brand must be stated on the individual product packaging.
 - 4.27.2. Inner cannula must connect securely to the stated tracheostomy tube to prevent accidental dislodgement.
 - 4.27.3. Reusable single patient use inner tubes must be able to be cleaned to prevent cross contamination and must have available cleaning / decontamination instructions and provide to NHS SC on request.
 - 4.27.4. Single patient use inner cannulas must be able to be inserted and reinserted multiple times and remain secure once reinserted.
 - 4.27.5. Information regarding the number of times an inner cannula can be removed, inserted and cleaned must be stated on the UOI/Individual packaging and/or IFUs.
- **4.28. Speciality Tracheostomy Tube** Speciality tracheostomy tubes are designed to maintain an airway for a variety of procedures. They include tracheostomy tubes such as extra proximal length tubes, flexible reinforced tubes.
- **4.29. Laryngectomy Tube** is a hollow device intended to maintain tracheostoma patency immediately after laryngectomy to provide an airway for the patient and to prevent tracheostomal stenosis (narrowing) in the months following the procedure. It is a soft plastic or silicone curved/contoured tube, similar to a tracheostomy tube, and may have an inner cannula and may be fenestrated to allow simultaneous use with a tracheoesophageal speech valve. It is held in place with a strap or band around the neck and is typically regularly removed and reinserted by the patient. This is a single patient use device. They must the following points:
 - 4.29.1. Laryngectomy tubes must have size stated on the tube to allow for identification.
 - 4.29.2. Size and length must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs.
 - 4.29.3. Laryngectomy tubes must be able to be cleaned to prevent cross contamination and must have available cleaning / decontamination instructions and provide to NHS SC on request.
 - 4.29.4. Must be supplied sterile.
- **4.30. Laryngectomy Inner Cannula** A sterile, hollow, cylindrical, curved device designed as an internal, independent component of a laryngectomy tube. It has precise

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 67 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

dimensions allowing it to fit exactly into the outer tube and function as the inner cannula for laryngectomy tubes that have this kind of function. It is removed once or more times daily to facilitate the removal of phlegm, slime, and contamination build-up, thereby preventing blockage and infection of the stoma and eliminates the traumatic removal of the laryngectomy tube There are a variety which can be single patient use.

- **4.31. Tracheostomy tube holders** are non-invasive devices designed to be fastened around the neck of a patient to secure a tracheostomy device (tube, button) in situ. It is typically in the form of a dedicated strap / collar intended to connect to the tracheostomy device.
 - 4.31.1. Device must be designed in such a manner to prevent airway trauma by tracheostomy tube movement whilst supporting the tracheostomy tube user to maintain movement as required. For example, touch fasteners, hook or ties.
 - 4.31.2. Device must minimise discomfort / potential harm to the patient through use of specialised materials or design for example.
 - 4.31.3. Must be Single patient use.
 - 4.31.4. Sizing information must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs.
 - 4.31.5. Must be adjustable for size to suit individual tracheostomy user.
 - 4.31.6. Must be able to be repositioned multiple times.
 - 4.31.7. Device must not cause damage to the tracheostomy tube through its use in accordance with manufacturer's instructions.
- **4.32. Tracheostomy Ties** Tracheostomy ties are the bands that go around the neck. They hold the tracheostomy tube in place. They must meet the following:
 - 4.32.1. Device must minimise discomfort / potential harm to the patient through use of specialised materials or design for example.
 - 4.32.2. Must be single patient use.
 - 4.32.3. Sizing information must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs.
 - 4.32.4. Must be adjustable for size to suit individual tracheostomy user.
- **4.33. Percutaneous Single Stage Dilation Set (PDT)** for placement of tracheostomy tube without direct surgical supervision of the trachea. Can contain surgical scalpel blade, an introducer needle, a guidewire, a small tracheal dilator, a protective sheath, a single-stage progressive tracheal dilator, a tracheal loading trocar, and a small slip-tip syringe. Must be supplied sterile.
 - 4.33.1. Can contain a tracheostomy tube in the kit, the size of the tracheostomy tube must be stated on individual product packaging.
 - 4.33.2. Training and implementation must be provided on request and in accordance with trust requirements through mutual agreement. This

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 68 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

could be for example, providing ongoing support including preimplementation education, supplier support and implementation guidance, bespoke clinical training, eLearning and post-implementation support.

- 4.33.3. Material type of product components must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
- **4.34. Tracheostomy Tube Accessories** which include replacement parts for the tracheostomy tube, equipment which connects to the tracheostomy tube to enable its operation excluding breathing systems products that offer the patient comfort during the use of the tube and products for maintaining the tracheostomy tube, for example:
 - products for measuring cuff pressure.
 - products for inflating/deflating the cuff.
 - Atomisation devices.
 - Cleaning Kits that contain as a minimum one cleaning brush and two disposable trays/bowls for cleaning liquid.
 - Plugs / Decannulation Cap.
 - Adaptors.
 - Speaking Valves.
 - Atomiser Spray Bottles.
 - Laryngectomy Accessories.
 - Dressings.
 - Bib Protectors.
 - Cleaning Brushes.
 - Artificial / Swedish Noses.

And sets to include more than one of the product lines contained within this Section 4.34 compiled in a convenient to use pack.

4.35. Supraglottic Devices - Supraglottic airways (SGAs) are a group of airway devices that can be inserted into the pharynx to allow ventilation, oxygenation, and administration of anaesthetic gases, without the need for endotracheal intubation. These devices are used for primary airway management, for rescue ventilation when facemask ventilation is difficult, and as a conduit for endotracheal intubation. They can be single use or reusable. Must be provided sterile.

4.36. Standards and Legislation – All Laryngeal masks and Supraglottic Devices Tubes and Connectors must conform to the below standards and legislation

STANDARD AND LEGISLATION	
BS ISO 11712:2009	
Anaesthetic and respiratory equipment. Supralaryngeal airways and connectors.	

4.37. Standard Laryngeal Masks - must meet the following:

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 69 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 4.37.1. Must have a standard 15mm male connector.
- 4.37.2. Material type must be stated on the individual product packaging and /or unit of issue packaging and/or IFUs and/or data sheets.
- 4.37.3. Must have an atraumatic tip to minimise risk of patient harm.
- 4.37.4. Cuff must have a smooth surface to minimise trauma to the patient.
- 4.37.5. The connector must not accidently detach when used in expected conditions in conjunction with the manufacturer's instructions.
- 4.37.6. The cuff must have a pilot balloon and / or other device such as a pressure indicator to indicate inflation / deflation of the cuff.
- 4.37.7. Must have a self-sealing valve for injecting air into the cuff.
- 4.37.8. The cuff be capable of being inflated and deflated multiple times.
- 4.37.9. The cuff must be able to be overinflated for pre-operative checks to ensure no herniations of the cuff or pilot balloon.
- 4.37.10. Volumes of air for the cuff inflation must be printed on the device.
- 4.37.11. The airway tube must be able to bend to a 180° angle with no kinking.
- 4.37.12. The IFU's must state compatibility of lubricant.
- **4.38. Reinforced Laryngeal Masks** have a wire coil inside the wall of the laryngeal mask to prevent inadvertent kinking of the device when in use. Must meet all requirements of paragraph 4.37 and the following:
 - 4.38.1. The metal wire must be embedded in the wall of the tube shaft.
 - 4.38.2. The wire reinforcement must be continuous along the length of the tube to minimise the risk of kinking.
- **4.39. Intubating Laryngeal Masks** The intubating laryngeal mask may be used for blind or fibreoptic bronchoscope-guided methods to aid intubation. Must meet all requirements of paragraph 4.37 and the following:
 - 4.39.1. Must have a circular channel to allow passage of an ET tube to support intubation in an airway rescue scenario.
 - 4.39.2. Compatibility with airway devices must be stated on the individual product packaging and /or unit of issue packaging and /or IFUs.
- **4.40. Preformed Laryngeal Masks** The preformed laryngeal mask has a preformed fixed anatomically curved structure to aid insertion. Must meet all requirements of paragraph 4.37.
- 4.41. Speciality Laryngeal Masks Speciality laryngeal masks include but are not limited to MRI safe devices, devices with gastric access or gastric blockage, video laryngeal masks. Must meet all requirements of paragraph 4.37.
- **4.42. Additional Supraglottic Devices** Additional supraglottic devices can incorporate but are not limited to, double cuffed tubes and double lumen tubes. Must meet all requirements of paragraph 4.37.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 70 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- **4.43. Non-Inflatable Cuffed Supraglottic Device** A device where the cuff is not required to be inflated and can be inserted into the airway where it forms to the patient's anatomy.
 - 4.43.1. Must have a standard 15mm male connector.
 - 4.43.2. Material type must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs and / or data sheets.
 - 4.43.3. Must have a circular channel to allow passage of an ET tube to support intubation in an airway rescue scenario.
 - 4.43.4. Must contain an integrated bite block.
 - 4.43.5. Must have an atraumatic tip to minimise risk of patient harm.
 - 4.43.6. Cuff must have a smooth surface to minimise trauma to the patient.
 - 4.43.7. If gastric channel is present, it must be designed in such a manner as to reduce risk of blockage and allow passage of an NG tube as required.
 - 4.43.8. The IFU's must state compatibility of lubricant.
- **4.44. Simple Airway Adjuncts** These generally consist of two groups of devices, oral airways and nasopharyngeal airways.
 - 4.44.1. Simple airway adjuncts are invaluable in increasing the success rate of basic airway manoeuvres, which aim to create and maintain airway patency, allowing spontaneous respiration or facilitating bag-mask ventilation.
- 4.45. Standards and Legislation All Oropharyngeal and Nasopharyngeal Airways, Tubes and Connectors must confirm to the below standards and legislation.

STANDARD AND LEGISLATION

BS EN ISO 5364:2016

Anaesthetic and respiratory equipment — Oropharyngeal airways

- **4.46. Oropharyngeal Airways** are curved metal or plastic tube inserted through the mouth to facilitate airway patency for gas exchange or suctioning. The device prevents the tongue from obstructing airflow. This can be a reusable device or single use. The most common are guedals or Berman airways and can include other types of oropharyngeal airways such as a Williams device:
 - 4.46.1. Must be individually packaged.
 - 4.46.2. Must have an atraumatic tip to minimise risk of patient harm.
 - 4.46.3. Must have a smooth exterior to minimise risk of patient harm and facilitate insertion.
 - 4.46.4. Material type must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
 - 4.46.5. Size and length must be stated on the individual product packaging and/or unit of issue packaging and / or IFUs.
 - 4.46.6. Must maintain its intended shape when removed from the manufacturers packaging.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 71 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 4.46.7. Tube material must have sufficient rigidity to have resistance to collapse and kinking.
- 4.46.8. Must be fully compliant to the standardised colour coding guidance outlined in ISO 5364:2016.
- 4.46.9. Must have a straight reinforced bite block to prevent occlusion of the air channel.
- 4.46.10. Must have a flange to prevent it falling back into the mouth.
- **4.47.** Nasopharyngeal Airways (NPA) are tubes designed to maintain airway patency from the nostril and the nasopharynx, bypassing the mouth and the oropharynx. The distal end is just above the epiglottis and below the base of the tongue. The proximal end of the device may have an integral 22mm connector for oxygen delivery. They are single use.
 - 4.47.1. Must have an atraumatic tip to minimise risk of patient harm.
 - 4.47.2. Must have a smooth exterior to minimise risk of patient harm and facilitate insertion.
 - 4.47.3. Material type must be stated on the individual product packaging and / or unit of issue packaging and / or IFUs and / or data sheets.
 - 4.47.4. Size and length must be stated on the individual product packaging and / or unit of issue packaging and/or IFUs.
 - 4.47.5. Must maintain its intended shape when removed from the manufacturers packaging.
 - 4.47.6. Tube material must have sufficient rigidity to have resistance to collapse and kinking.
 - 4.47.7. The size must be marked on the outside of the tube.
 - 4.47.8. Must have a flange integral to the device to limit depth of insertion.
- **4.48. Supraglottic and Airway Adjuncts Accessories** includes equipment that directly connects to the airway device to enable its operation, products for measuring and inflating cuffs, atomisation devices, bite guards for protection during product use, products that directly assist in the placement, use or removal of the mask; and sets which include more than one of the relevant product lines contained within this Framework compiled in convenient to use packs.

5. Lot 3- Breathing System Circuits and Accessories

5.1. This Lot is for the supply of parts that make up the anaesthetic breathing systems / circuits and the critical care breathing systems / circuits. These systems attach to universal and specified anaesthetic machines, universal and specified ventilator machines as well as universal and specified flow drivers used within operating rooms, recovery, emergency and intensive care environments. These parts will include the components which make up the different breathing systems / circuits.

The anaesthesia breathing circuit is a non-sterile assembly of devices designed to conduct medical gases from the fresh gas supply outlet of an anaesthesia unit/workstation to the patient typically connecting the patient, a ventilator / ventilation bag, carbon dioxide (CO2)

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 7:		Page 72 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

absorber, and a monitor. It may be conductive or non-conductive. This can be reusable device or single use.

Please note, Breathing Systems for non-invasive ventilation are not within the specification of this tender as there is a separate NHS SC Framework for this area.

A **non-rebreathe system** is an anaesthesia breathing system from which all the expired mixture is discharged.

A **partial rebreathing system** is an anaesthesia breathing system in which a portion of the expired mixture is retained in the system.

5.2. Breathing Systems within this Lot include but are not limited to:

- 5.2.1. Coaxial Breathing Systems.
- 5.2.2. Coaxial Breathing Systems with a Monitoring Line.
- 5.2.3. Coaxial Breathing Systems with a Monitoring Line and 2L Bag.
- 5.2.4. Coaxial Breathing Systems with a Flow Sensor.
- 5.2.5. Coaxial Breathing Systems with a Flow Sensor and Expiratory Valve.
- 5.2.6. Circle Breathing Systems.
- 5.2.7. Circle Breathing Systems with a Monitoring Line.
- 5.2.8. Circle Breathing Systems with a Monitoring Line and 2L Bag.
- 5.2.9. Circle Breathing Systems with a Flow Sensor.
- 5.2.10. Circle Breathing Systems with a Flow Sensor and Expiratory Valve.
- 5.2.11. Breathing Systems for Dedicated Machine.
- 5.2.12. Parallel Y (WYE) Breathing Systems without Rebreathing Bags and Bagging Limb.
- 5.2.13. Parallel Y (WYE) Breathing Systems with Bagging Limb.
- 5.2.14. Parallel Y (WYE) Breathing Systems with Bag and Limb.
- 5.2.15. Parallel Y (WYE) Breathing Systems with Bag and Limb, and Specialist Connectors.
- 5.2.16. Humidified Parallel Y (WYE) Breathing Systems.
- 5.2.17. Breathing Systems Antimicrobial.
- 5.2.18. Mapleson A Commonly used during anaesthesia.
- 5.2.19. Mapleson B Commonly used during recovery and emergency situations.
- 5.2.20. Mapleson C Commonly used during recovery and emergency situations.
- 5.2.21. Mapleson D Commonly used during anaesthesia.
- 5.2.22. Mapleson E Commonly used during anaesthesia.
- 5.2.23. Mapleson F Jackson-Rees Modification.
- 5.2.24. Mapleson F with open tail bag.
- 5.2.25. Mapleson F with closed tail bag and APL valve.
- 5.2.26. Single Limb Breathing Systems.
- 5.2.27. Single Limb Breathing Systems with Heated Wire.
- 5.2.28. Transport Breathing Systems both dedicated machine and non-machine dedicated can be made of additional parts can include monitoring lines, filters, non-rebreathing valves, elbows, flow sensor tubing.

5.3. Standards and Legislation

STANDARD AND LEGISLATION		
LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 73 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Where components are included in the Breathing system they must conform to the following standards, where applicable.

BS EN ISO 23328-2:2009

Breathing system filters for anaesthetic and respiratory use. Non-filtration aspects.

BS EN ISO 23328-1:2008

Breathing system filters for anaesthetic and respiratory use. Salt test method to assess filtration performance.

HMEs must conform to:

International standard ISO 9360-1:2009

Anaesthetic and respiratory equipment. Heat and moisture exchangers (HMEs) for humidifying respired gases in humans. HMEs for use with minimal tidal volumes of 250 ml.

5.4. Breathing Systems and their components - All breathing systems must:

- 5.4.1. Must be robust, compact, and light weight.
- 5.4.2. Must have the ability to remove waste gases, effectively eliminate Co2.
- 5.4.3. Must have ability to warm and humidify inspired gas.
- 5.4.4. Able to deliver intended gas mixture.
- 5.4.5. Must permit spontaneous, controlled, or assisted ventilation in all age groups.
- 5.4.6. Be designed to protect patients from barotrauma.
- 5.4.7. Have low resistance: Should have minimal length, maximal internal diameter and be without sharp curves or sudden changes in diameter
- 5.4.8. Have minimal dead space within the circuit.
- 5.4.9. All breathing system components that have a requirement for a port with a cap: the cap must be tethered to the device.
- 5.4.10. Material type must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
- 5.4.11. Where a product is reusable, product must be able to be cleaned to prevent cross contamination and must have available cleaning / decontamination instructions must be stated on the individual product packaging or IFU and/or made available to NHS Supply Chain or end user on request.
- 5.4.12. Maximum working pressure must be made available for breathing system.
- **5.5. Breathing Tubes** are non-rigid tubes used to convey gases and/or vapours between components of a breathing system they can be they must:
 - 5.5.1. Tubing must have flexibility and resist kinking
 - 5.5.2. Must have ability to act as reservoir.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 74 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 5.5.3. Must have some distensibility but not enough to prevent excessive pressures from developing in the circuit.
- 5.5.4. Must be lightweight and have low resistance.
- 5.5.5. Internal diameter 22mm of tubing must be labelled on packaging and/or within IFU's.
- 5.5.6. The resistance to the flow of gas must be listed within the IFU's and/or made available.
- 5.5.7. Must be stated on individual product packaging type of tubing e.g. if coaxial, corrugated or smooth bore.
- **5.6. Reservoir Bag** a collapsible gas container which is a component within a breathing system and must:
 - 5.6.1. Material type must be stated on the individual product packaging and/or unit of issue packaging and/or IFUs and/or data sheets.
 - 5.6.2. Must have size stated on individual product packaging and/or unit of issue packaging.
 - 5.6.3. Must be able to contain reservoir of gas without absorption.
 - 5.6.4. Must have airtight seal when connected to breathing system.
 - 5.6.5. Must be able to squeeze reservoir bag for assisted or controlled ventilation.
- 5.7. Adjustable Pressure Limiting Valve also known as APL valve pop-off valve
 - 5.7.1. Must allow flow of waste gas and fresh gas to leave the breathing system.
 - 5.7.2. Provide pressure control within breathing system.
- **5.8.** Accessories for use on all anaesthesia and intensive care breathing systems and include all the additional connectors and parts that connect to the breathing system, accessories within this Lot include:
 - 5.8.1. Catheter Mounts.
 - 5.8.2. Pressure Monitoring Lines.

6. Lot 4 - Humidification and Filtration (HME's, HMEF's, Bacterial and Viral Filters)

6.1. Humidification and Filtration devices are to provide warmth, moisture and filtration manually. These devices and filters do not require the need for heated humidification circuit or cold humidification circuit and are used independently. Breathing Filters for use on all anaesthesia and intensive care breathing systems. These include the breathing filters that protect the patient, the breathing system and the equipment from bacteria and viruses. The heat and moisture exchangers (HME) that conserve expired heat and moisture and return it to the patient, as well as the combined HME and Filter options (HMEF) Filters. Breathing Filters, HME's and HMEF's can be supplied with or without a CO₂ port. This Lot includes:

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 75 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 6.1.1. Electrostatic Filters without HME.
- 6.1.2. Electrostatic Filters with HME (HMEF).
- 6.1.3. Pleated / Mechanical Filters without HME.
- 6.1.4. Pleated / Mechanical Filters with HME (HMEF).

6.2. Standards and Legislation

STANDARD AND LEGISLATION

BS EN ISO 23328-2:2009

Breathing system filters for anaesthetic and respiratory use. Non-filtration aspects.

BS EN ISO 23328-1:2008

Breathing system filters for anaesthetic and respiratory use. Salt test method to assess filtration performance.

BS EN ISO 9360-1:2009

Anaesthetic and respiratory equipment. Heat and moisture exchangers (HMEs) for humidifying respired gases in humans - HMEs for use with minimal tidal volumes of 250 ml.

- 6.3. All devices within this Lot must meet the following:
 - 6.3.1. Suppliers must be able to provide following information to NHS supply chain if required: recommended maximum duration of use, internal volume, hydrophobic filter, HME has tethered cap, Tidal volumes, if product is latex free, sterility.
 - 6.3.2. Must have tethered cap if cap present on device.
 - 6.3.3. Must have flow resistance of product and pressure at 30L pre/post conditioning (cmH₂O).
 - 6.3.4. Must have flow resistance of product and pressure at 60L pre/post conditioning (cmH₂O).
- **6.4. Breathing Mechanical / Electrostatic Filters** are devices intended to reduce transmission of particulates, including micro-organisms, such as bacteria and viruses to prevent cross infection to and from the patient during anaesthesia or other types of ventilation. Must meet the following:

- **6.5. Heat Moisture Exchange (HME)** conserve heat and moisture during expiration and make this available to inspired gases during subsequent inspiration. They must meet the following:
 - 6.5.1. Must state the internal volume in millimetres on product packaging or in IFU's.
 - 6.5.2. Must be tested against the international standard ISO 9360-1:2000(E).

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 76 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

^{6.4.1.} Filter filtration performance rate must meet ISO 23328-1 (Salt test).

- 6.5.3. Must state the recommended range of tidal volumes and / or flowrates on IFU's or product packaging.
- 6.5.4. Must state the gas leakage in millimetres per minute on product packaging or IFU's.
- **6.6. Heat Moisture Exchange Filters** HMEF's are a combination of an HME and Breathing Filter to achieve both clinical outcomes of filtration and heat and moisture exchange.
 - 6.6.1. Ergonomic shape with no sharp edges reduces pressure marking
 - 6.6.2. Must have flow resistance of product and pressure at 30L pre/post conditioning (cmH₂O).
 - 6.6.3. Must have flow resistance of product and pressure at 60L pre/post conditioning (cmH₂O).

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	evision: 7 August 2017 Page 77 of 90	
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

5(b) Tender Response Document

Non-Financial Questions (Technical Envelope)

Question	Description	Response
1.1.1	(*) How much stock will you hold (for NHS Supply Chain) for each Product Line	
	which is ready for immediate dispatch in the UK?	
1.1.2	(*) In the event of receiving unforeseen demand beyond your planned stock	
	capabilities, what would be your maximum lead time to get stock into your UK/EU	
	distribution centre across all Products submitted as part of this Lot?	
1.1.3	(*) For all Products submitted as part of this Lot, should they be listed as E-direct, what will be your longest lead time for this route of delivery to either the customer	
	or NHS Supply Chain?	
1.1.4	(*) For all Products submitted as part of this Lot, should they be listed as Blue	
	Diamond, what will be your longest lead time for this route of delivery to either the	
	customer or NHS Supply Chain?	
1.2.1	(*) If you are successfully awarded to this Framework Agreement, within what	
	period of time will you make an initial response to any product complaints raised by	
	NHS Supply Chain or directly by NHS customers?	
1.2.2	(*) If you are successfully awarded to this Framework Agreement, when will your	
1.2.3	Customer Service Help Desk/Support be available to respond to an end user?	
1.2.3	(*) What timeframe will you respond to general customer or NHS Supply Chain queries (such as queries regarding products, practice, service etc.)?	
1.3.1	(*) If a customer requests additional training, for example following a switch from a	
1.5.1	different supplier, will you provide a training and support package that is available	
	face to face, online, downloadable, or other format?	
1.3.2	(*) Do you take into consideration feedback from patients and carers when	
	developing products?	
1.3.3	(*) "Will you provide a dedicated clinical contact within your organisation to	
	support the Framework Agreement in delivering requested training and answering	
	any questions from NHS Trusts, organisations or NHS Supply Chain?	
	(A Clinical Contact must be a nurse or AUD registered with the environment	
	(A Clinical Contact must be a nurse or AHP registered with the appropriate professional body)	
1.4.1	(*) To increase the transparency of supply chains to the NHS, please can you outline	
	the % products applicable to this Framework Agreement which you will have	
	mapped the supply chain for to the most significant levels of production by	
	commencement date of the Framework Agreement?	
.4.2	(*) The NHS England Public Board approved a roadmap to help suppliers align with	
	NHS' net zero ambition between now and 2030. In line with the NHS net zero	
	roadmap, and with a focus on carbon oversight and reduction, please can you	
	outline the % of products applicable to this Framework Agreement that you will have a carbon footprint for by 2028?	
1.4.3	(*) The Plastic Packaging Tax came into force in the UK on 1st April 2022. It applies	
	at a rate of £200/tonne on plastic packaging with less than 30% recycled plastic,	
	manufactured or imported into the UK (including packaging on goods which are	
	imported).	
	With a focus on the reduction of carbon for the NHS, and as a minimum to address	
	specification point 2.2 "External product packaging must be made of recyclable	
	material where possible", please outline the % of products applicable to this	
	Framework Agreement that will have a minimum of 30% recycled plastic across all	
4.4	packaging content for the commencement date of the Framework Agreement?	
.4.4	(*) NHS Supply Chain's experiences through COVID-19 highlighted the criticality of ensuring a resilient supply chain for NHS trusts, through our delivery networks and	
	suppliers' supply chains. The NHS has evolving service need and requires continued	
	support to deliver safe and excellent patient care. To ensure this continuity of	
	supply to the NHS and its patients, does your Organisation have a supply resilience	
	management system in place?	
	Note: A supply resilience management system specifies the practices and	
	procedures for ensuring supply continuity, mitigating against disruptive events, and	
	response to supply disruption incidents.	
1.4.5	(*) Continuing the focus on supply chain resilience, does your Organisation have a	
	Continuity Plan and Emergency Response Procedure in place?	

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 78 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Awarded Lines [To be inserted as part of the final Framework Agreement] Awarded Additional Lines [To be inserted as part of the final Framework Agreement] Discounts & Savings [To be inserted as part of the final Framework Agreement]

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 79 of		Page 79 of 90

NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)

Schedule 6

Commercial Schedule

Part 1

- The Parties acknowledge that the Contract Price is the basis upon which the Framework Agreement is awarded and unless amended in accordance with this Schedule 6 and/or Schedule 7 (as the case may be) the Contract Price shall remain fixed during the Term. For the avoidance of doubt the Supplier shall not be entitled to unilaterally adjust the Contract Price.
- 2. Management Fee The Management Fee does not apply to all Orders for Goods placed pursuant to this Agreement in accordance with Clause 9 of Schedule 2.
- During the Term of the Framework Agreement each Party may approach the other to discuss special offers, discounts, value added offerings and commitment or bulk buy deals. Neither Party shall be obliged to accept any offer made by the other.
- 4. The Supplier agrees to work with NHS Supply Chain during the Term of the Framework Agreement to identify cost saving opportunities, including the way in which Goods are sourced, supplied, ordered and packaged, which can be reflected in a more competitive Contract Price.
- If the Supplier requests an increase to the Contract Price it must provide justification to NHS Supply Chain for such increase and NHS Supply Chain may in its absolute discretion consent to such increase.
- 6. To drive greater transparency and stability in contract pricing NHS Supply Chain reserves the right to review various cost price mechanisms throughout the Term of the Framework Agreement. In the event that there is a fluctuation in commodity costs used in the production of related finished goods that could impact the contracted price, NHS Supply Chain reserves the right to review the Contract Price in conjunction with the Supplier if evidenced through a commodity index tracker mechanism agreed with both parties.
- 7. Once a price variation has been agreed by both Parties pursuant to this Schedule 6 the new Contract Price shall take effect on the date agreed by the Parties in writing.

Part 2

1. Additional and Associated Goods and Services

- 1.1. Without limitation to the provisions of Clause 22.1 of Schedule 2, the Supplier acknowledges to NHS Supply Chain that over the Term, additional goods and services may be made available for purchase under the Framework Agreement. Such additional goods and services may also include the provision of associated goods, materials or items associated with those additional goods and services (which together shall be "the Additional and Associated Goods and Services").
- 1.2. Additional and Associated Goods and Services to NHS Supply Chain will be made available for purchase under the Framework Agreement at the sole discretion of NHS Supply Chain. In order to determine whether the Additional and Associated Goods

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 80 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

and Services will be made available for purchase under the Framework Agreement NHS Supply Chain shall consider a number of different factors, including (but not limited to) whether the proposed Additional Solutions and Associated Goods are deemed to be within scope of the procurement exercise under which the Framework Agreement was awarded.

- 1.3. The Supplier acknowledges and agrees that, to the extent relevant, any Additional and Associated Goods and Services must comply with the standards set out in the Specification and the Tender Response Document.
- 1.4. Without prejudice to any of the other provisions set out in Schedule 2, NHS Supply Chain reserves the right to undertake at its absolute discretion a review of the Goods and Services which are supplied under the Framework Agreement. Following such review, NHS Supply Chain may change supply routes for any of the Goods and Services and/or remove certain Goods and Services and/or Additional and Associated Goods and Services from the Framework Agreement.

2. Price Saving Initiatives

- 2.1. NHS Supply Chain reserves the right to share savings information in order to assist Authorities with making informed procurement decisions.
- 2.2. Such savings initiatives shall include (but shall not be limited to):
 - 2.2.1. The publication of price ranking sheets showing the Supplier's ranking as to price for a particular Good(s);
 - 2.2.2. Data arising from the Compare and Save Programme;
 - 2.2.3. Re-opening of competition in accordance with Schedule 7 for the supply of certain Goods and/or listing in the NHS Supply Chain Catalogue through a particular route of supply for a specified period of time;
- 2.3. Notwithstanding the provisions of Schedule 7 NHS Supply Chain recognises that:
 - 2.3.1. the pricing set out in Schedule 5(b) Discounts and Savings may be used where commitment can be gained from an Authority and without reopening competition;
 - 2.3.2. during the lifetime of the Framework Agreement, the Supplier may want to offer additional savings to a Participating Authority (or group of Participating Authorities) through the provision of discounted pricing, value added offerings, commitment, bulk buy initiatives, direct rebates (i.e. payments which are agreed directly between the Supplier and a Participating Authority or groups of Participating Authorities for Orders which are placed under the Framework Agreement) occasional special offers, (for instance in relation to new product introductions) NHS year-end spend and market growth incentives.
 - 2.3.3. NHS Supply Chain may request pricing on behalf of a Participating Authority or group of Participating Authorities in return for commitment by NHS Supply Chain or a Participating Authority or group of Participating Authorities to

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7 August 2017 Page 81 of 90		
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

purchase an agreed value and/or volume of Goods that may or may not be for an agreed period of time. In these instances NHS Supply Chain may request and agree improved or different terms from those which are set out in the Framework Agreement (and pass these to the Participating Authority or group of Participating Authorities (as the case may be) for their consideration), including terms in relation to the matters set out below:

- i. the Contract Price in respect of some or all of the Goods;
- ii. the quantity of Goods which shall be ordered and whether the Goods subject to any Order shall be delivered in a single or multiple instalments;
- iii. whether the Goods subject to an Order shall be delivered to NHS Supply Chain, or delivered directly to the relevant Participating Authority;
- iv. whether the Supplier shall be required to monitor the volume of Goods stored at any Participating Authority's premises and to deliver further Goods in accordance with any instructions agreed with or given by NHS Supply Chain or the relevant Participating Authority;
- the time at which the Supplier may issue its invoice in respect of any Goods subject to an Order, and whether such invoice shall be paid by NHS Supply Chain or the relevant Participating Authority;
- vi. the transfer of risk in and title to any Goods subject to an Order.

3. Compare and Save Programme

- 3.1. NHS Supply Chain may, during the lifetime of the Framework Agreement, undertake a 'Compare & Save' (or similar) programme aimed at updating an alternative database for goods which have been awarded across this Framework Agreement.
- 3.2. Alternatives may include both like for like products and products which require a change of practice, training or switch of existing equipment by the Authority but which ultimately offer the same output. There may also be the option for the Supplier to add comments regarding specific information about the product i.e. training required, additional products required, only compatible with a certain machine. Suggested comparable products will be reviewed by NHS Supply Chain before they are made available to an Authority. Where the Supplier updates an alternative with another supplier's product NHS Supply Chain will automatically reverse this so that the other product is also detailed as an alternative.
- 3.3. In order to ensure this comparable database contains accurate like for like goods, NHS Supply Chain may from time to time request support from the Supplier in order to verify and update the data which is stored on this database.
- 3.4. The comparable products will be:
 - 3.4.1. Used to suggest alternatives to Authorities in the event of a stock out situation;

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	evision: 7 August 2017 Page 82 of 90	
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 3.4.2. Used to record savings from Authorities who are switching products to achieve savings this information will then be reported to Supply Chain Coordination Limited; and
- 3.4.3. accessed by NHS Supply Chain staff and shared with Authorities at the Authorities' request.
- 3.5. Without prejudice to the provisions of paragraph 3.1 above, NHS Supply Chain reserves the right, in order to support Authorities on their savings initiatives, to actively market the information and inform Authorities of the savings opportunities through the Compare and Save programme

4. Supply Chain Simplification

- 4.1. NHS Supply Chain reserves the right to engage with the Supplier with regard to implementing supply chain simplification initiatives in order to assist in reducing the cost of Goods throughout the lifetime of the Framework Agreement. The supply chain simplification initiatives will aim to remove complexity and cost from the inbound supply chain from the point of manufacture through to delivery to NHS Supply Chain warehouses. Examples of such initiatives include, but are not exclusive to:
- 4.2. Packaging optimisation (i.e. cubic volume reduction, component elimination, material Specification reduction, etc.);
- Logistics/transport optimisation (i.e. pallet fill increase, vehicle/container fill increase, etc.);
- 4.4. Direct inbound delivery (i.e. supply of product into NHS Supply Chain Distribution Centres on a containerised basis directly from overseas manufacturing facilities, bypassing UK / EU warehousing).

5. Incoterms

5.1. In addition to the provisions mentioned at paragraph 4 above (Supply Chain Simplification) NHS Supply Chain reserves the right to engage with the Supplier on various Incoterms (as more particularly set out in the ORS) throughout the lifetime of the Framework Agreement.

6. Samples

- 6.1. From time to time, NHS Supply Chain may request samples of the Goods to be provided to such location (as may be reasonably required by NHS Supply Chain) to a Participating Authority. Such samples shall be provided Free of Charge and may be used to inform a Participating Authority's purchasing decision under the Framework Agreement.
- 6.2. Without prejudice to the provisions set out in paragraph 6.1 above NHS Supply Chain may (on behalf of the Clinical Evaluation Team ("CET")) request samples to be provided by the Supplier (on a free of charge basis) over the lifetime of the Framework Agreement. Such samples shall be assessed by CET in order to determine whether such Goods shall be included in the reopening of competition in accordance with the terms of this Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017 Page 83 of 90	
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

6.3. Where such samples are requested in accordance with paragraph 5.2 above they must be provided to a "Ward Ready Standard" and which shall mean that such samples shall be provided in the packaging, box and Unit of Issue as a Participating Authority would expect to see at ward level. Failure to provide samples to a Ward Ready Standard or in accordance with the timescales which may reasonably be required by NHS Supply Chain may result in the Supplier being excluded from any reopening of competition in accordance with is held in accordance with the terms of the Framework Agreement.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 84 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 7

Ordering Procedure and Order Form

1. Contracts based on Framework Agreement without amendments

- 1.1. NHS Supply Chain may elect to purchase Goods and/or Services from such supplier(s) on the Framework Agreement as it may at its discretion choose, on the terms and at the Contract Price as calculated in accordance with Schedule 6.
- 1.2. As set out at paragraph 1.1 above, NHS Supply Chain may place an Order for Goods and/or Services on the supplier(s) based on the terms of this Framework Agreement, including the Call-off Terms and Conditions for the Supply of Goods, at any time during the Term. Such Order(s) shall form a Contract between the supplier(s) and NHS Supply Chain which shall comprise the following documents:
 - 1.2.1. the Call-off Terms and Conditions for the Supply of Goods;
 - 1.2.2. a completed Order Form;
 - 1.2.3. the applicable parts of the Specification and Tender Response Document set out at Schedule 5 of this Framework Agreement, as may be supplemented by information set out and/or referred to in the Order Form; and
 - 1.2.4. any relevant provisions applicable to the Contract as set out in the Framework Agreement.

2. Contracts based on Framework Agreement with amendments

- 2.1. Alternatively, where the Tender Response Document in relation to the Framework Agreement has insufficient detail to be able to price up bespoke requirements and award a Contract without further information, NHS Supply Chain may invite either: (i) one capable supplier; or (ii) some capable suppliers; or (iii) all capable suppliers (which in each case may not include the supplier) to submit an offer in relation to (a) a specific requirement of a Participating Authority. NHS Supply Chain reserves the right to reopen competition using any of (1) an eAuction; (2) threshold pricing; and (3) target pricing or other methodologies advised to suppliers.
- 2.2. Where more than one supplier is invited to submit an offer, the offers shall be evaluated using such criteria as a Participating Authority shall determine (including (i) price only; (ii) quality only; or (iii) a combination of quality and price) and in each case, the terms of the resulting Contract (including the Contract Price and Specification of the relevant Goods) may differ from those set out in the Framework Agreement and Call-Off Terms and Conditions.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 85 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

- 2.3. As set out at paragraphs 2.1 2.2 above, NHS Supply Chain may at the request of one or more Participating Authorities re-open competition between the supplier(s) referred to at paragraph 2.1 above based on the award criteria referred to at paragraph 2.2 above. Once a competition has been concluded Orders may be placed by NHS Supply Chain under the Call-off Terms and Conditions for the Supply of Goods (as amended in accordance with the competition). A Contract concluded following a re-opening of competition is made up of the following components:
 - 2.3.1. the bespoke requirements referred to in paragraph 2.1 above;
 - 2.3.2. the Call-off Terms and Conditions for the Supply of Goods (as amended in accordance with the competition);
 - 2.3.3. a completed Order Form;
 - 2.3.4. the applicable parts of the Specification and Tender Response Document set out at Schedule 5 of this Framework Agreement, as may be supplemented by information used to conduct the competition; and
 - 2.3.5. any relevant provisions applicable to the Contract as set out in the Framework Agreement.
- 2.4. For the avoidance of doubt, any competition pursuant to this Framework Agreement shall be carried out by NHS Supply Chain only on behalf of one or more Participating Authorities. No Participating Authority (other than NHS Supply Chain) shall be entitled to carry out a competition under this Framework Agreement without the express consent of NHS Supply Chain.

3. General

In relation to either or both of paragraphs 1 and 2 above, NHS Supply Chain may request pricing on the basis of a commitment by a Participating Authority to purchase a specified volume or value of Goods during an agreed period of time, for which NHS Supply Chain may pay the supplier in advance.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 86 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 8

Service Levels

1. The Supplier agrees to conform to the following KPIs during the Term of this Framework Agreement:

On time Deliveries on time subject to a tolerance of +/- thirty (30) minutes (for the avoidance of doubt, deliveries which arrive on time but are not unloaded due to the driver's decision, deliveries which do not arrive and deliveries which arrive at the wrong delivery location shall also be considered late)	98%
Quantity Quantity of delivery correct against the relevant Order (including deliveries in excess and shortfall of the Order quantity)	98%
Quality Quality of delivery in accordance with the Framework Agreement and Contracts (including delivery presentation in accordance with the Framework Agreement and Contracts (the delivery must be presented in such a way that it can be unloaded safely and in a ready for use condition taking into consideration the Framework Agreement and Contract requirements) and damaged Goods (the Goods must be in a condition that is new and ready to use))	95%
Administration Timely and accurate administration (including booking/amending delivery times and Orders and invoices, delivery advice notes and labels being in accordance with the requirements of the Framework Agreement and Contracts)	99%

- 2. Any KPI discrepancy attributable to an act or omission of NHS Supply Chain (or another Participating Authority) shall not be used to calculate the Supplier's sub-standard performance level.
- A service level shall be generated for each of the KPIs in relation all Orders placed on the Supplier within each calendar month during the Term of the Framework Agreement and a monthly average service level for each KPI shall be calculated ("Monthly Service Level").
- 4. The Supplier's performance shall be measured:
 - 4.1. in relation to Orders placed pursuant to a Non-direct Route of Supply by NHS Supply Chain; and
 - 4.2. in relation to Orders placed pursuant to a Direct Route of Supply by the Authority and the Supplier.
- 5. In relation to Orders placed pursuant to a Direct Route of Supply the Supplier shall submit monthly reports to NHS Supply Chain outlining its performance in relation to the KPIs for the preceding month. Such report shall be submitted to NHS Supply Chain not later than the 14th day of the month following the month to which the report relates. NHS Supply Chain may verify the information provided by the Supplier with the Authority and reserves

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 87 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

the right to amend the Supplier's monthly service level report in accordance with the findings of such verification exercise.

- 6. Should the Monthly Service Level of the Supplier fall below the relevant KPI:
 - 6.1. on two (2) or more occasions in any six (6) month period in relation to On time and/or Quantity; and
 - 6.2. on three (3) or more occasions in any six (6) month period in relation to Quality and/or Administration,

NHS Supply Chain may serve a performance notice on the Supplier. The Supplier shall present to NHS Supply Chain within thirty (30) days of receipt of such performance notice an action plan to improve the Supplier's Monthly Service Level ("Action Plan"). The Parties shall, within ten (10) Business Days of NHS Supply Chain receiving the Action Plan meet to discuss and agree the Action Plan. NHS Supply Chain may make reasonable amendments to the Action Plan to improve the Supplier's performance. The Action Plan must include a timetable for improvement of the Supplier's performance to, as a minimum, the level required by Clause 1 of this Schedule 8 in relation to the relevant KPI. Such timetable shall be agreed by the Parties but shall in any event be no longer than six (6) months.

- 7. In the event that the Supplier:
 - 7.1. fails to produce an Action Plan in accordance with Clause 6 of this Schedule 8; or
 - 7.2. fails to improve its Monthly Service Level to the minimum level required by Clause 1 of this Schedule 8 within the timetable set out in the Action Plan in accordance with Clause 6 of this Schedule 8,

the Supplier shall be considered to have committed a material breach capable of remedy for the purpose of Clause 16.3 of Schedule 2.

- 8. Notwithstanding Clauses 6 and 7 of this Schedule 8, where the Monthly Service Level in relation to Quantity, Quality or On time of Goods delivered against the relevant Order(s) falls below the relevant KPI, NHS Supply Chain shall (without prejudice to its rights to claim for any other categories of loss arising from such failure to meet the relevant KPI) be entitled to raise a debit note to the Supplier for a sum equal to the loss NHS Supply Chain has incurred or suffered in relation to lost margin and the cost, if any, of purchasing alternative goods and/or services (and any related administrative costs) as a result of the shortfall in ready to use delivery quantity against the relevant Orders. The Parties agree this is a true and fair assessment of loss suffered through lost margin and the cost of purchasing alternative goods and/or services (and any related administrative costs). Where NHS Supply Chain does not elect to raise a debit note in the manor detailed in this paragraph, then it shall remain entitled to claim damages.
- 9. If the Supplier disputes NHS Supply Chain's Monthly Service Level as applicable to the Supplier, the Supplier shall provide evidence to NHS Supply Chain that the Monthly Service Level is incorrect within seven (7) days of disputing such Monthly Service Level and the Parties shall meet to discuss any necessary amendment to the Monthly Service Level. If the Parties cannot agree the Monthly Service Level the matter shall be referred to the dispute resolution procedure set out in Clause 23 of Schedule 2.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 88 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

10. For the avoidance of doubt, nothing in this Schedule 8 shall limit in any way either Party's rights and remedies, including the right to claim damages and or termination rights which may arise, under this Framework Agreement or any Contract.

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods		
Revision: 7	August 2017	Page 89 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)		

Schedule 9

Call-Off Terms and Conditions for the Supply of Goods

LEGAL TEMP 810-01 NHS Supply Chain Framework Agreement for the Supply of Goods			
Revision: 7		August 2017	Page 90 of 90
NHS Supply Chain Operated by Supply Chain Coordination Ltd (SCCL) (registered number 10881715)			